ARCHAEOLOGY ACT **DECLARATION RE**

Under authority of section 19 of the Archaeology Act Stats. P.E.I. 2006, c. 2 Council ordered that a Proclamation do issue proclaiming the said "Archaeology Act" to come into force effective 4 April 2009.

EC2009-141

ARCHAEOLOGY ACT REGULATIONS

Pursuant to section 17 of the Archaeology Act R.S.P.E.I. 1988, Cap. A-17.1, Council made the following regulations:

1. In these regulations

Definitions

- (a) "Act" means the Archaeology Act R.S.P.E.I. 1988, Cap. A-17.1;
- (b) "designation" means the designation by the Minister of an designation archaeological site as a protected archaeological site under subsection 4(1) of the Act;
- (c) "municipality" means the municipality in which an municipality archaeological site is located;
- (d) "notice of intention to designate an archaeological site as a notice of intention protected archaeological site" means a notice of the Minister's archaeological site intention to make an order designating an archaeological site as a as a protected protected archaeological site under clause 4(2)(a) of the Act;

to designate an

- (e) "owner" includes a person with a legal interest in the parcel of owner land containing an archaeological site;
- (f) "registry" means an office of the Registrar as defined in the registry Registry Act R.S.P.E.I. 1988, Cap. R-10;
- (g) "temporarily protected archaeological site" means an temporarily archaeological site that is, by order of the Minister, designated as a protected temporarily protected archaeological site under subsection 4(4) of

archaeological site

2. The Minister shall apply the following criteria in determining whether to make an order designating an archaeological site as protected designation archaeological site under subsection 4(1) of the Act:

- (a) the age of the site;
- (b) the significance of the site;
- (c) the context of the site;
- (d) the integrity of the site;
- (e) the contribution or potential for contribution to the understanding of the history of the province.
- 3. (1) For the purposes of clause 4(2)(b) of the Act and these regulations, Prescribed persons the following persons are the prescribed persons:

- (a) a person who is the owner of the parcel of land containing an archaeological site:
- (b) the municipality in which an archaeological site is located;
- (c) a person who owns a parcel of land adjoining an archaeological site.
- (2) Where the Minister intends to make an order designating an Notice of Intention archaeological site as a protected archaeological site, the Minister shall cause a copy of the notice of intention to designate an archaeological site as a protected archaeological site, in Form 1, as set out in the Schedule to these regulations,

- (a) to be served on the prescribed persons in respect of the archaeological site; and
- (b) to be published in the Gazette.
- (3) A notice of intention to designate an archaeological site as a Contents of notice protected archaeological site shall contain
 - of intention to designate (a) the municipal address and a legal description of the
 - archaeological site; (b) a summary of the significance of the proposed designation order;
 - (c) information respecting the consequences of the proposed designation order:
 - (d) a date and time by which the prescribed persons or an interested or affected party may comment on or object to the proposed designation order;
 - (e) information in respect of how to make an objection to the proposed designation order; and
 - (f) such other information as the Minister considers necessary or advisable

(4) The Minister

(a) may amend a notice of intention to designate an archaeological site as a protected archaeological site, in Form 2, as set in the Schedule to these regulations; and

(b) shall serve and publish the amended notice of intention to designate an archaeological site as a protected archaeological site in accordance with subsection (2).

Amended notice of

- (5) The Minister may cancel a notice of intention to designate an Notice of archaeological site as a protected archaeological site at any time before
- the designation occurs.

- (6) Where the Minister cancels a notice of intention to designate an Idem archaeological site as a protected archaeological site under subsection (5), the notice of cancellation of the notice of intention to designate an archaeological site as a protected archaeological site shall be in Form 3, as set out in the Schedule to these regulations.
- **4.** No person who is served with a notice of intention to designate an archaeological site as a protected archaeological site shall, within 120 days after the date the notice of intention to designate has been served and published in accordance with subsection 3(2), alter, destroy or undertake any work in respect of the archaeological site that is proposed to be designated as a protected archaeological site.

No alteration for 120 days

under subsection 3(2); or

- Notice of Objection (a) prescribed person who is served with a notice of intention to designate an archaeological site as a protected archaeological site
- (b) interested or affected party likely to be effected by a proposed designation order.

may object to the proposed designation order by filing a notice of objection to a protected archaeological site designation, in Form 4, as set out in the Schedule to these regulations, with the Minister within 30 days after the date of the publication in the Gazette of the notice of intention to designate an archaeological site as a protected archaeological site.

6. (1) Where no notice of objection to a protected archaeological site Designation if no objection designation is filed with the Minister in accordance with section 5, the

Minister may make an order under subsection 4(1) of the Act designating an archaeological site as a protected archaeological site.

(2) Where a notice of objection to a protected archaeological site Consideration of designation is filed with the Minister in accordance with section 5, the Minister, within 90 days after the date of service of the notice of objection,

objections

- (a) shall consider every such notice of objection filed; and
- (b) may
 - (i) designate the archaeological site as a protected archaeological site as described in the notice of intention to designate an archaeological site as a protected archaeological site, or
 - (ii) cancel the notice of intention to designate an archaeological site as a protected archaeological site.
- (3) Where the Minister designates an archaeological site as a protected Service of notice archaeological site, the Minister shall cause a copy of the notice of protected archaeological site designation, in Form 5, as set out in the Schedule to these regulations,
 - (a) to be served on the prescribed persons and any interested or affected party who filed a notice of objection to a protected archaeological site designation;
 - (b) to be published in the Gazette; and
 - (c) to be filed in the Register and appropriate registry.
- 7. (1) Where the Minister has, by order, designated an archaeological site

 Temporary designation

 Temporary designation as a temporarily protected archaeological site under subsection 4(4) of the Act, the Minister shall cause a copy of the notice of temporarily protected archaeological site designation, in Form 6, as set out in the Schedule to these regulations,

- (a) to be served on the prescribed persons;
- (b) to be published in the Gazette; and
- (c) to be filed in the Register and appropriate registry.
- (2) The notice of temporarily protected archaeological site designation Contents of notice shall contain

- (a) the municipal address and a legal description of the archaeological site;
- (b) the specified period for which the archaeological site has been designated as a temporarily protected archaeological site;
- (c) a summary of the penalties for a contravention of the notice of temporarily protected archaeological site designation;
- (d) the date and time by which the prescribed persons may comment, in writing, on the notice of temporarily protected archaeological site designation: and
- (e) such other information as the Minister considers necessary or advisable.
- (3) The Minister may cancel a notice of temporarily protected Cancellation of archaeological site designation order by issuing a notice of cancellation designation designation of temporary protected archaeological site designation, in Form 7, as set out in the Schedule to these regulations, and such cancellation shall be served, published and filed in accordance with the manner set out in subsection (1).

8. (1) The owner of the parcel of land containing an archaeological site Appeal from may appeal a protected archaeological site designation order made by the Minister under subsection 4(1) of the Act, by filing a notice of appeal with the Commission, not later than 30 days after the date the protected archaeological site designation order was made by the Minister.

- (2) A notice of appeal filed with the Commission under subsection (1) Service of notice of shall be in the form approved by the Commission and shall be served on:
 - (a) the Commission;
 - (b) the Minister.
- (3) The Commission shall give written notice to the owner of the Notice of appeal parcel of land containing the archaeological site, the Minister, and any other interested or affected party of the date, time and place of the hearing of the appeal.

(4) The Commission shall, when hearing an appeal, determine its own Commissioner to procedure, subject to adherence to the rules of natural justice.

determine its own procedure

(5) The Commission shall hear and decide an appeal and shall issue an Decision of the order giving effect to its disposition.

Commission

(6) The Commission may vary, confirm or cancel the protected Cancel order archaeological site designation order.

(7) The Commission shall give written reasons for its decision and Certified copy of provide a certified copy of its decision or order to the appellant and the decision or order Minister.

9. (1) The Minister may cancel an order designating an archaeological Cancellation of site as a protected archaeological site where the Minister is satisfied the archaeological site has been damaged or destroyed, or as the Minister considers necessary or advisable.

(2) Where the Minister, under subsection (1), has cancelled an order of Notice published, designation of an archaeological site as a protected archaeological site, the Minister shall cause a copy of the notice of cancellation of an archaeological site as a protected archaeological site designation, in Form 8, as set out in the Schedule to these regulations,

served, filed

- (a) to be served on the prescribed persons;
- (b) to be published in the Gazette; and
- (c) to be filed in the Register and appropriate registry.
- (3) The cancellation of an archaeological site as a protected Does not affect archaeological site designation order does not affect an investigation of, or charges brought against, any person for an offence committed under the Act or these regulations.

investigation,

(4) An archaeological site that had been designated as a protected May be rearchaeological site and that was subject to a designation order which has designated been cancelled, may be re-designated as a protected archaeological site in accordance with the procedure for designation set out in the Act.

10. (1) An application for an archaeological permit under subsection Applicant 7(1) of the Act shall be in Form 9, as set out in the Schedule to these requirements regulations, and include the following information:

- (a) a statement that the applicant will accept
 - (i) responsibility for all of the scientific aspects of the archaeological investigation both in the field and in the laboratory, and
 - (ii) responsibility for the conduct of the archaeological investigation in the field;
- (b) a current resume including the applicant's educational background as well as any experience pertinent to the proposed archaeological investigation;
- (c) documentation evidencing that the applicant has received supervised field training in archaeological field surveying and excavation and has demonstrated the capacity to design and carry out the project described in the application for an archaeological permit;
- (d) where an archaeological site is likely to be of significance to the aboriginal community, information in respect of any consultations that have taken place with the aboriginal community.
- (2) The holder of an archaeological permit issued by the Minister under Recording of subsection 7(2) of the Act shall ensure that all archaeological site information is properly recorded on such forms as the Minister may approve.

archaeological site information

(3) An archaeological permit issued by the Minister under subsection Archaeological 7(2) of the Act shall be in Form 10, as set out in the Schedule to these permit regulations.

(4) Where the Minister is satisfied that an application for an Significance to archaeological permit involves an archaeological site that is, or is likely to be, of significance to the aboriginal community, the Minister shall refuse to issue the archaeological permit unless the Minister is satisfied that appropriate consultations have taken place with such aboriginal groups as the Minister considers appropriate.

aboriginal

11. An application for a work permit under subsection 8(1) of the Act Application for shall be in Form 11, as set out in the Schedule to these regulations, and work permit include the following information:

- (a) a description of the proposed work the applicant intends to carry out at the protected archaeological site;
- (b) a statement explaining why the work is necessary at the protected archaeological site;
- (c) the proposed timeline in respect of the schedule of work to be carried out at the protected archaeological site;
- (d) the names of the architect and contractor conducting the work at the protected archaeological site, including their contact information;
- (e) a statement explaining how the work will impact the protected archaeological site;
- (f) proposed mitigation measures to be carried out at the protected archaeological site.
- 12. (1) An archaeological permit or a work permit is valid only for the validity of permit specified time, description of work to be carried out, and locations of work to be carried out at a protected archaeological site that are outlined in the permit.

(2) A person who holds an archaeological permit or a work permit Permit produced to shall produce the permit to the Director, upon request.

inspector

13. (1) A person who holds an archaeological permit or a work permit Amendment of may apply to the Minister to amend the permit.

(2) The Minister may allow, refuse, or vary the requested amendment Amendment of in respect of an archaeological permit or a work permit in whole or in part, and the amended permit shall be re-issued to specify any such amendment or variation.

14. (1) The holder of an archaeological permit or a work permit shall, within 60 days after the expiration of such permit or within such further period as is specified in the permit, submit to the Minister, in writing, details in respect of the locations of all newly discovered archaeological sites and information in respect of all revisits to known archaeological sites.

(2) The holder of an archaeological permit or a work permit shall, Project report within one year after the expiration of such permit or within such further period of time as is specified in the permit, submit a final project report to the Minister in accordance with subsection (3).

(3) The final project report submitted to the Minister by the holder of Project report an archaeological permit or a work permit shall

- (a) describe the fieldwork undertaken and the location and extent of the investigations at each archaeological site;
- (b) describe the methodologies used in data acquisition, recording and analysis including field, archival and laboratory investigations;
- (c) describe the environmental factors and cultural history relevant to the investigations at each archaeological site;
- (d) assess the current physical status of the archaeological site and any present or potential factors that could alter the current physical status of the site;
- (e) interpret the significance and age of the archaeological site investigated, based on a summary investigation of the findings; and
- (f) assess the results of the archaeological site investigation in relation to the scope and objectives of the project, as described in the archaeological permit or work permit.
- (4) The holder of an archaeological permit or a work permit shall Publications or submit to the Minister copies of all publications or theses produced by the permit holder that contain any references to archaeological research conducted under the archaeological permit or work permit.

15. These regulations come into force on April 4, 2009.

Commencement

SCHEDULE

SCHEDULE

FORM 1 NOTICE OF INTENTION TO DESIGNATE AN ARCHAEOLOGICAL SITE AS A PROTECTED ARCHAEOLOGICAL SITE

TAKE NOTICE THAT I, the Minister of Communities, Cultural Affairs and Labour for the Province of Prince Edward Island, intend to designate the following archaeological site as a protected archaeological site, as described in the attached legal description, in accordance with the Act and the regulations:	
(street	or road name and number)
	(municipality)
(property number)	(owner's name)
significant archaeological feature in including all land appurtenant to it,	th the Act and the regulations, this site represents a Prince Edward Island and I intend to designate this site, as a protected archaeological site on or after to be protected under the authority of the Act.
undertake works that may affect any	ected archaeological site, no person shall build on or y site designated under the Act without a permit. By the

authority of section 4 of the regulations, no person shall change or alter the site or area for a period of 120 days from the date of service of this notice, unless this notice is otherwise cancelled under the Act or the regulations.

Any prescribed person or interested or affected party likely to be effected by the intended designation order may object to the intended designation by serving a notice of objection to a protected archaeological site designation within 30 days of the date of the publication of this Notice in the Gazette. Service of a notice of objection to a protected archaeological site designation must be made on me, in care of:

Archaeology Coordinator
Division of Culture, Heritage, and Libraries
Department of Communities, Cultural Affairs and Labour
P.O. Box 2000, Charlottetown, PE C1A 7N8.

A notice of objection to a protected archaeological site designation form may be obtained from that office.

If a notice of objection to a protected archaeological site designation is served upon me within the prescribed time, I am required to proceed in accordance with subsection 6(2) of the regulations. If no notice of objection to a protected archaeological site designation is served as prescribed, I may proceed with the designation in accordance with subsection 6(1) of the regulations.

•	
DATED at Charlottetown, this	day of,
	Minister of Communities, Cultural Affairs and Labour

FORM 2 ARCHAEOLOGY ACT AMENDED NOTICE OF INTENTION TO DESIGNATE AN ARCHAEOLICAL SITE AS A PROTECTED ARCHAEOLOGICAL SITE

archaeological site dated	designate an archaeological site as a protected was served and published in
	ulations with respect to the following site:
(street	or road name and number)
	(municipality)
(property number)	(owner's name)
	d it necessary that the notice of intention to designate an chaeological site be amended as set out below.
protected archaeological site is amer	f intention to designate an archaeological site as a nded to read as follows:
DATED at Charlottetown, this	day of,
	Minister of Communities, Cultural Affairs and Labour
	FORM 3 N OF NOTICE OF INTENTION TO DESIGNATE E AS A PROTECTED ARCHAEOLOGICAL SITE
archaeological site datedAct and the regulations with respect	
(street	or road name and number)
	(municipality)
(property number)	(owner's name)
	elled the said notice of intention to designate an chaeological site, pursuant to section 9 of the
DATED at Charlottetown, this	day of,

Minister of Communities, Cultural Affairs and Labour

FORM 4 ARCHAEOLOGY ACT NOTICE OF OBJECTION TO A PROTECTED ARCHAEOLOGICAL SITE DESIGNATION

WHEREAS a notice of intention to designate an archaeological site as a protected archaeological site dated
(street or road name and number)
(municipality)
(property number) (owner's name)
TAKE NOTICE THAT I object to the proposed archaeological site designation, pursuant to section 5 of the regulations.
OBJECTOR: Name:
Address:
Telephone: Fax Status of Objector: owner of the parcel of land containing the archaeological site lessee of the parcel of land containing the archaeological site affected person by a proposed designation order
interested person to a proposed designation order
Reasons for objection:
(Signature of Objector)
(Date)
(Title, if applicable)
FORM 5 ARCHAEOLOGY ACT NOTICE OF PROTECTED ARCHAEOLOGICAL SITE DESIGNATION WHEREAS a notice of intention to designate an archaeological site as a protected archaeological site dated
accordance with the Act and the regulations with respect to the following site:
(street or road name and number)
(municipality)
(property number) (owner's name)
AND WHEREAS
no notice of objection to a protected archaeological site designation was served in accordance with the Act and the regulations; or the Minister considered the objections filed and ordered the designation pursuant to section 6 of the regulations; or the Island Regulatory and Appeals Commission confirmed or varied the scope of the designation pursuant to section 8 of the regulations.
I HEREBY DESIGNATE the said site as a protected archaeological site pursuant to section 6 of the regulations.
The owner of the parcel of land containing the archaeological site may appeal the designation order by serving written notice, in the approved form, to the Island Regulatory and Appeals Commission, within 30 days of service of this Notice of protected archaeological site designation.
DATED at Charlottetown, this day of

FORM 6 ARCHAEOLOGY ACT NOTICE OF TEMPORARILY PROTECTED ARCHAEOLOGICAL SITE

	DESIGNATION
Province of Prince Edward Island, the following site, as described in t and the regulations:	ster of Communities, Cultural Affairs and Labour for the temporarily designate as a protected archaeological site, he attached legal description, in accordance with the Act
(stree	et or road name and number)
	(municipality)
(property number)	(owner's name) in
for the period fromaccordance with clause 7(2)(b) of t	to in he regulations.
	ats a significant archaeological site, and temporarily for the purposes of further investigation respecting
or area on the said property for a pe	shall damage, disturb, excavate or otherwise alter the site eriod of 45 days from the date of service or publication unless this notice is otherwise cancelled.
designation order is an offence for	benalty for contravention of the conditions of a temporary which a fine for an individual may be up to \$10,000 and in accordance with section 16 of the Act.
effected by this notice may comme	prescribed person or any interested or affected party ent, in writing, to the Minister, in respect of this notice of e 4 p.m. within five business days of the issuance of this
DATED at Charlottetown, this	day of,
	Minister of Communities, Cultural Affairs and Labou
NOTICE OF CANCELI	FORM 7 RCHAEOLOGY ACT ATION OF TEMPORARILY PROTECTED LOGICAL SITE DESIGNATION
	y protected archaeological site designation dated and published pursuant to section 7 of the regulations
	et or road name and number)
	(municipality)
(property number)	(owner's name)
	celled the above described notice of temporarily nation on this date, pursuant to subsection 7(3) of the
DATED at Charlottetown, this	day of,
	Minister of Communities, Cultural Affairs and Labou

FORM 8 ARCHAEOLOGY ACT NOTICE OF CANCELLATION OF PROTECTED ARCHAEOLOGICAL SITE DESIGNATION

WHEREAS ona protected archaeological site pursuant to subs	section 4(1) of the Act:
(street or road na	me and number)
(munici	pality)
(property number)	(owner's name)
TAKE NOTICE THAT I hereby cancel the pro- above described site, as of this date, pursuant to	
DATED at Charlottetown, this day of	of,
	of Communities, Cultural Affairs and Labour
FOR ARCHAEOI APPLICATION FOR AN ARC	LOGY ACT
I (surname, first name)of (address, telephone number, e-mail address)	
representing (institution)	
materials and how these requirements will (g) Significance of proposed project (h) Relation of project to previous work (i) Description of how archaeological site (j) Expected completion date (i.e. date of	poordinates and civic address location) oject ogy ion requirements for recovered artifactual ll be met or other work in progress will be restored final report) cated for the preservation of archaeological
2. Plan for deposition of artifacts, materials an	d samples that may be completed:
3. Sponsoring Agencies:	
4. Previous permit(s) awarded applicant:	
5. Personnel (attach curriculum vitae for all pr personnel):	incipal researchers and supervisory
(Attach additional sheet if required)	
I certify that I am familiar with the provisions of and regulations, and that I will abide by and cothe regulations, and will abide by the terms and	mply with the requirements of the Act and
Date: Place: .	
	(Signature of applicant)

FORM 10 ARCHAEOLOGY ACT PERMIT TO CONDUCT ARCHAEOLOGICAL INVESTIGATIONS

This permit is issued in accordance with and subject to the provisions of the Archaeology Act and the regulations made under the Act. 1. Issued to: 2. Official responsible: 3. Location(s) for which permit is valid: 4. Valid for period: 5. Limitations and restrictions: (b) (c) Issued by me pursuant to subsection 7(2) of the Archaeology Act. Date issued: Signed: Minister of Communities, Cultural Affairs and Labour FORM 11 ARCHAEOLOGY ACT APPLICATION FOR A WORK PERMIT the Minister of Communities, Cultural WHEREAS on Affairs and Labour designated as a protected archaeological site the following site: (street or road name and number) (municipality) (property number) (owner's name) AND WHEREAS the owner of the protected archaeological site wishes to commence work to change, alter, move or disturb the site as set out herein: APPLICANT: Name Address Fax If the Applicant was not the owner of the archaeological site at the time of the designation, give information respecting the deed to the Applicant: County Liber Folio Folio Date of registration of deed to Applicant: Nature of project and description of the proposed work: How will this work effect the protected archaeological site? What mitigation measures are proposed to reduce the impact on the protected archaeological site? Reasons for the proposed work: Estimated project cost: .. Sources and amount of funding: Projected start date:

Projected end date:

	ork projected for this site?
(Note: separate permits v Architect/Engineer for project:	vill be required for future work)
Address:	
Phone: General Contractor for proposed work: Address:	Fax:
Phone:	Fax: applied for not required
	(Signature)
	(Date)
	(Title, if applicable)
ARCHAEOLOGICAL	ORM 12 EOLOGY ACT IMPACT STATEMENT (AIS) rsuant to subsection 5(3) of the Act)
THIS Archaeological Impact Statement (A	alS) is filed respecting the following site:
`	d name and number)
(mu	unicipality)
(property number)	(owner's name)
Address	
If the Applicant was not the owner of the partner the time of the designation, give information	Fax
HIS prepared by:	
Address	
Telephone	Fax
	ttach curriculum vitae)
Detailed description of heritage impacts id	entified:
Detailed description of results of AIS:	
Recommendations (including cost estimate	es):
I certify that I have made this AIS after con	assessment to the best of my qualifications and
	(Signature)
	(Date)

FORM 13 ARCHAEOLOGY ACT STOP ORDER

	(Issued by the Minister pursual	nt to subsection 5(1) of the Act)
To:		
IN RESI	PECT OF:	
	Activity [Description
TAKE N	ОТІСЕ ТНАТ:	
	you are engaged in an activity likely ogical site:	to result in damage to or destruction of an
	to subsection 5(1) of the Archaeolog ted below:	ry Act you are hereby directed to suspend work
	and are to remain under suspension	be suspended effective, until further directed in writing by the
	O)R
suspension	ollowing portions of the above Projection date stated below and are to remainly the Minister.	et are to be suspended effective on the n under suspension until further directed in
	WORK AFFECTED	SUSPENSION DATE
	to take place	owing investigations, excavations or other
This Stop	o Order is issued on	by:
	Minister	r of Communities, Cultural Affairs and Labour
Acknowl	ledgement of receipt of the Stop Ord	er on, 20
Name (P	rint)	Signature
		Capacity

EXPLANATORY NOTES

SECTION 1 is the definition section.

SECTION 2 outlines the criteria to be applied by the Minister in determining whether to make an order designating an archaeological site as a protected archaeological site.

SECTION 3 defines the persons who are the prescribed persons for the purposes of the Act and these regulations. The provision also outlines the requirements for and contents of a notice of intention to designate an archaeological site as a protected archaeological site.

SECTION 4 prohibits the alteration or destruction of an archaeological site for a period of 120 days after the date of a notice of intention to designate an archaeological site as a protected archaeological site has been served and published.

SECTION 5 outlines who may file a notice of objection in respect of an archaeological site that is proposed to be designated as a protected archaeological site, as well as the requirements for filing a notice of objection to the proposed designation.

SECTION 6 outlines what the Minister may do in circumstances in which either a notice of objection in respect of an archaeological site that is proposed to be designated as a protected archaeological site is filed or where no notice of objection is filed.

SECTION 7 provides for the temporary designation of an archaeological site as a temporarily protected archaeological site and outlines the circumstances in which the Minister may cancel a temporary designation order.

SECTION 8 provides for an appeal to the Island Regulatory and Appeals Commission by the owner of the parcel of land containing an archaeological site in respect of the designation of an archaeological site as a protected archaeological site.

SECTION 9 authorizes the Minister to cancel an order designating an archaeological site as a protected archaeological site in the described circumstances.

SECTION 10 outlines the requirements for an application for an archaeological permit and requires consultation with aboriginal groups, where an archaeological site is likely to be of significance to the aboriginal community.

SECTION 11 outlines the requirements for an application for a work permit.

SECTION 12 outlines the circumstances in which an archaeological permit or a work permit is valid, and requires the holder of such permit to produce the permit to the Director, upon request.

SECTION 13 provides for the amendment of an archaeological permit or a work permit.

SECTION 14 outlines the requirements for the submission by an archaeological permit holder or a work permit holder, after the expiry of such permit, of a final project report that is to be submitted to the Minister in respect of newly discovered archaeological sites. The final project report is to include information in respect of all revisits to known archaeological sites.

SECTION 15 provides for the commencement of these regulations.

FINANCIAL ADMINISTRATION ACT CERTAIN ACCOUNTS RECEIVABLE UNDER THE ENVIRONMENT TAX ACT AUTHORITY TO WRITE-OFF

Pursuant to subsection 26.1(1) of the Financial Administration Act R.S.P.E.I. 1988, Cap. F-9 Council authorized the write-off of certain accounts receivable under the Environment Tax Act totalling \$211.42 (including interest to 28 February 2009) as follows:

SCHEDULE (ENVIRONMENT TAX ACT, WRITE-OFFS)

ACCOUNT OF

AMOUNT WRITTEN-OFF

Blue, Michael sometimes carrying on business under the style name Blue's Automotive & Performance Montague

\$211.42

EC2009-143

FINANCIAL ADMINISTRATION ACT CERTAIN ACCOUNTS RECEIVABLE UNDER THE GASOLINE TAX ACT **AUTHORITY TO WRITE-OFF**

Pursuant to subsection 26.1(1) of the Financial Administration Act R.S.P.E.I. 1988, Cap. F-9 Council authorized the write-off of certain accounts receivable under the Gasoline Tax Act totalling \$3,425.15 (including interest to 28 February 2009) as follows:

SCHEDULE (GASOLINE TAX ACT, WRITE-OFFS)

ACCOUNT OF

AMOUNT WRITTEN-OFF

J.D. Johnston Investments Ltd. sometimes carrying on business under the style name Johnston Transport Summerside	he	\$ 1,760.93
MacDougall, Roger Cardigan		816.80
Millbrook Steel Corp. Summerside		176.06
Polar Foods International Morell		104.76
Williams, William H. Tyne Valley		<u>566.60</u>
Tyne vancy	Total	\$ <u>3,425.15</u>

FINANCIAL ADMINISTRATION ACT CERTAIN ACCOUNTS RECEIVABLE UNDER THE REVENUE TAX ACT **AUTHORITY TO WRITE-OFF**

Pursuant to subsection 26.1(1) of the Financial Administration Act R.S.P.E.I. 1988, Cap. F-9 Council authorized the write-off of certain accounts receivable under the Revenue Tax Act totalling \$308,900.14 (including interest to 28 February 2009) as follows:

SCHEDULE (REVENUE TAX ACT, WRITE-OFFS)

ACCOUNT OF	AMOUNT WRITTEN-OFF
Anchor Lock & Safe Inc. Charlottetown	\$ 9,646.84
Anderson, Adam Charlottetown	126.93
Arsenault, Maynard Charlottetown	158.64
Bellefontaine, Lindsey sometimes carrying on business under the style name LuLu Belle's Charlottetown	382.78
Blue, Michael sometimes carrying on business under the style name Blue's Automotive & Performance Montague	10,134.26
Bradley, Peter D. sometimes carrying on business under the style name West Covehead Collision Repair West Covehead	1,218.85
Campbell, David sometimes carrying on business under the style name J & A Welding Belle River	818.56
Cornell Trading Ltd. sometimes carrying on business under the style name La Cache Montreal, QC	910.41
Curran, Gail sometimes carrying on business under the style name Peeks & Perks Art Gallery & Cafe Summerside	4,769.63
Curran, John L. Charlottetown	\$334.24
Curran, Patrick Mount Stewart	\$185.53

EXECUTIVE COUNCIL	26 MARCH 2009

DesRoches, Vance Summerside		318.50
Flynn, Diane St. Peters		104.39
Gallant, Beth sometimes carrying on business under the style name Brookside Cottage South Rustico		1,147.51
Groom Foods Inc. Summerside		150,647.55
Hammoran Inc. sometimes carrying on business under the style name Mid-Isle Mitsubishi Charlottetown		2,783.09
Henry, Percey J. Georgetown		234.42
Howatt, Stephen Cornwall		746.82
Jamieson, Donald Montague		162.70
Lewis, Allison York		167.13
Linens 'N Things Canada Corp. New Jersey		56,534.07
McCannell, Dave sometimes carrying on business under the style name Aerials by Dave Stratford		1,335.89
O'Grady, Jerry sometimes carrying on business under the style name O'Grady's Upholstery North Wiltshire		1,705.68
Patstone Holdings Ltd. sometimes carrying on business under the style name Teddy Bear Makers Saint John, NB		3,348.37
Seabrook Farms Limited Summerside		568.12
Seafury Industries Limited Summerside		44,527.58
The Flower Cart Ltd. Charlottetown		<u>15,881.65</u>
	Total	\$ <u>308,900.14</u>

HIGHWAY TRAFFIC ACT COMMERCIAL VEHICLE (TRIP INSPECTION AND RECORDS) REGULATIONS AMENDMENT

Pursuant to subsection 148(1) of the Highway Traffic Act R.S.P.E.I. 1988, Cap. H-5, Council made the following regulations:

- 1. Section 1 of the Highway Traffic Act Commercial Vehicle (Trip Inspection and Records) Regulations (EC225/89) is amended
 - (a) in clause (b), by the addition of the words "means a commercial vehicle" after the words "commercial vehicle"; and
 - (b) by the addition of the following after clause (c):
 - (c.1) "major defect" means a major defect listed in Column 3 of major defect Schedule I, II or III or in Column 2 of Schedule IV;
 - (c.2) "minor defect" means a minor defect listed in Column 2 of minor defect Schedule I, II or III;
 - (c.3) "motor coach" means a bus of monocoque design, designed to $\,$ $_{motor\,coach}$ provide intercity, suburban, commuter or charter service and equipped with air ride suspension, air brakes, automatic brake adjusters and under-floor baggage storage;

- 2. Section 2 of the regulations is amended by the deletion of the words "This Part does not apply to" and the substitution of the words "The following vehicles are exempt from the inspection requirements outlined in these regulations:".
- 3. Subsection 3(1) of the regulations is amended by the deletion of the word "his" and the substitution of the words "the driver's".
- 4. Sections 4 and 5 of the regulations are amended by the deletion of the words "Schedules I and II" and the substitution of the words "Schedules I to IV".
- 5. The regulations are amended by the addition of the following after section 5:
- 5.1 (1) Schedules I to IV apply to the inspection of the following Inspection commercial vehicles:

- (a) Schedule I applies to the daily inspection of trucks, tractors and trailers drawn by a truck or tractor;
- (b) Schedule II applies
 - (i) to the daily inspection of buses designed, constructed and used for the transportation of passengers with a designated seating capacity of more than 10, including the driver, but excluding its operation for personal use, and
 - (ii) to any trailer towed by a bus;
- (c) Schedule III applies to the daily inspection of motor coaches;
- (d) Schedule IV applies to the under-vehicle inspection of a motor coach, which is valid for 30 days or 12,000 kilometres, whichever occurs first.
- (2) Every person who conducts a daily inspection of a commercial Inspection of vehicle or trailer shall ensure that the inspection includes an inspection of every system and component listed in Column 1 of the appropriate daily inspection schedule.

(3) Every person who conducts a daily inspection of a commercial Minor or major vehicle or trailer shall ensure that the inspection is adequate to determine whether there is a minor defect or a major defect as set out in Column 2 or Column 3, as the case may be, of the appropriate daily inspection schedule.

(4) A daily inspection of a commercial vehicle or trailer is valid for 24 Validity of hours

(5) A daily inspection of a motor coach conducted in accordance with Under-vehicle Schedule III is not valid unless an under-vehicle inspection of the motor inspection coach has been conducted in accordance with Schedule IV, and the under-vehicle inspection is still valid.

- (6) Every person who conducts an under-vehicle inspection of a motor *Idem* coach shall ensure that the under-vehicle inspection includes an inspection of every system and component listed in Column 1 of Schedule IV.
- (7) Every person who conducts an under-vehicle inspection of a motor Idem coach shall ensure that the under-vehicle inspection is adequate to determine whether there is a major defect as set out in Column 2 of Schedule IV
- (8) Every person who detects a defect in an under-vehicle inspection of Idem a motor coach shall ensure that the defect is repaired before the motor coach is driven or operated.
- (9) Every defect detected in an under-vehicle inspection of a motor Idem coach must be repaired before the motor coach is driven or operated in order for the under-vehicle inspection to be considered complete.
 - (10) An under-vehicle inspection of a motor coach

Idem

- (a) must be conducted while the vehicle is positioned over a pit or raised in a manner that provides adequate access to all of the applicable vehicle components; and
- (b) must be conducted by a person who holds the technician certification or qualification required in the province.
- (11) An under-vehicle inspection of a motor coach is valid

Idem

- (a) until the 31st day after the date the under-vehicle inspection is conducted: or
- (b) until the motor coach has been driven 12,000 kilometres, whichever occurs first

6. The heading immediately before section 10 of the regulations and sections 7 to 10 of the regulations are revoked and the following substituted:

PART II INSPECTION REPORTS

7. (1) A person conducting an inspection of a commercial vehicle in Report to be accordance with Schedule I, II or III shall prepare an inspection report, in completed writing, or in an equivalent electronic format, that contains the following information:

- (a) the licence plate, VIN or unit number(s) of the vehicle(s);
- (b) the carrier's name;
- (c) the date(s) and time(s) of inspection;
- (d) the city, town, village or highway location where the inspection was performed:
- (e) a statement signed by the person conducting the inspection and by the person driving the vehicle (if different than the person inspecting the vehicle) that the vehicle(s) identified on the report has (have) been inspected in accordance with the applicable requirements;
- (f) the legible printed name of the person(s) conducting the inspection:
- (g) the signature(s) of the person(s) conducting the inspection;
- (h) the odometer reading(s) (if equipped).
- (2) A person conducting an under-vehicle inspection of a motor coach Idem in accordance with Schedule IV shall prepare an inspection report, in writing, or in an equivalent electronic format, that contains the following information.
 - (a) the licence plate, VIN or unit number(s) of the vehicle(s);
 - (b) the carrier's name;
 - (c) the date(s) and time(s) of inspection;
 - (d) the location(s) where the inspection was performed;
 - (e) a statement signed by the person conducting the inspection and by the person driving the vehicle (if different than the person inspecting the vehicle) that the vehicle(s) identified on the report has (have) been inspected in accordance with the applicable Schedule IV requirements;
 - (f) the legible printed name of the person(s) conducting the inspection;
 - (g) the signature(s) of the person(s) conducting the inspection;
 - (h) the odometer reading(s) (if equipped).

(3) No carrier shall permit a person to drive a commercial vehicle, and Prohibition no person shall drive a commercial vehicle on a highway unless the vehicle is inspected, as required.

(4) No carrier shall permit a person to drive a commercial vehicle, and Idem no person shall drive a commercial vehicle on a highway unless that person is in possession of the required inspection reports and applicable Schedules.

(5) A driver of a commercial vehicle shall provide a paper or an Inspection reports equivalent electronic copy of the required inspection reports and applicable Schedules on the demand of an inspector.

8. (1) Where no defects are detected during an inspection of a where no defects commercial vehicle, the person conducting the inspection shall record are detected that fact on the inspection report.

(2) A person conducting an inspection of a commercial vehicle in When defects are accordance with Schedule I, II or III shall record on the inspection report detected any defects detected during the inspection and shall report such defects to the carrier or a person appointed by the carrier prior to the next required inspection.

(3) A person conducting an under-vehicle inspection of a motor coach Information to be in accordance with Schedule IV shall record brake adjustment measurements, all defects detected during the inspection, and the nature of all repairs that are carried out on the motor coach.

inspections

(4) The driver of a commercial vehicle shall monitor the vehicle's Driver to monitor condition in accordance with the schedule of inspection items listed in Column 1 of Schedules I to IV, and when defects are detected by the driver, the driver shall record the defects on an inspection report and report the defects to the carrier prior to the next required inspection.

driving

(5) Where major defects are detected by the driver of a commercial Major defects to vehicle while the vehicle is in operation, such defects shall be recorded on an inspection report and be immediately reported to the carrier by the driver of the commercial vehicle.

reported immediately

(6) No carrier shall permit a person to drive a commercial vehicle, and Vehicle not to be no person shall drive a commercial vehicle on a highway when a major defect is present on the vehicle.

operated with major

(7) Every carrier shall ensure that all previously reported commercial Carrier to ensure vehicle defects are rectified prior to the next required inspection or within defect are corrected a timeframe specified by the jurisdiction of travel.

9. (1) The driver of a commercial vehicle shall provide the original copy Report to be given of each inspection report to the carrier who is responsible for the to carrier commercial vehicle within 20 calendar days after the date each inspection report is completed.

(2) Every carrier shall retain the original copy of each inspection report Carrier records and certification of repairs for at least six months after the date each inspection report is completed.

10. The requirements on a driver of a commercial vehicle or the person Exemption from conducting an inspection of a commercial vehicle to prepare inspection preparation of reports for short reports under these regulations do not apply to the driver of a truck, trips tractor and trailer when the truck, tractor and trailer is being operated within a radius of 160 kilometres from the location at which the driver reported for work if the driver is to return to that location and be released from work.

7. Section 11 of the regulations is amended

- (a) in clause (a), by the deletion of the words "section 10" and the substitution of the words "sections 7 and 8"; and
- (b) by the revocation of clause (b) and the substitution of the following:
- (b) the inspection reports referred to in subsections 7(1) and (2), as completed by the driver or person conducting the inspection if different than the driver, for a period of three months from the dates on which the inspection reports were completed.

8. Section 12 of the regulations is amended by the deletion of the word "his" and the substitution of the words "the driver's".

- 9. Schedules I and II of the regulations are revoked and Schedules I to IV as set out in the Schedule to these regulations are substituted.
- 10. These regulations come into force on April 4, 2009.

Schedule

Schedule I – Daily Inspection of Trucks, Tractors and Trailers

 $\label{eq:Application:} Application:$ This Schedule applies to trucks, tractors and trailers or combinations thereof exceeding a registered gross vehicle weight of 4500 kg.

Column 1	Column 2	Column 3
Systems and	Minor Defect(s)	Major Defect(s)
components		
Part 1. Air Brake System	Audible air leak. Slow air pressure build-up rate.	Pushrod stroke of any brake exceeds the adjustment limit. Air loss rate exceeds prescribed limit. Inoperative towing vehicle (tractor) protection system. Low air warning system fails or system is activated. Inoperative service, parking
Part 2. Cab	Occupant compartment door	or emergency brake. • Any door fails to close
Part 3. Cargo Securement	fails to open. • Insecure or improper load covering.	Insecure cargo. Absence, failure, malfunction or deterioration of required cargo securement device or load covering.
Part 4. Coupling Devices	Coupler or mounting has loose or missing fastener.	Coupler is insecure or movement exceeds prescribed limit. Coupling or locking mechanism is damaged or fails to lock. Defective, incorrect or missing safety chain or cable.
Part 5. Dangerous Goods		Dangerous goods requirements not met.
Part 6. Driver Controls	Accelerator pedal, clutch, gauges, audible and visual indicators or instruments fail to function properly.	
Part 7. Driver Seat	Seat is damaged or fails to remain in set position.	Seatbelt or tether belt is insecure, missing or malfunctions.
Part 8. Electric Brake System	Loose or insecure wiring or electrical connection.	Inoperative breakaway device. Inoperative brake.
Part 9. Emergency Equipment and Safety Devices	• Emergency equipment is missing, damaged or defective.	,,
Part 10. Exhaust System	• Exhaust leak.	Leak that causes exhaust gas to enter the occupant compartment.
Part 11. Frame and Cargo Body	Damaged frame or cargo body.	• Visibly shifted, cracked, collapsing or sagging frame member(s).
Part 12. Fuel System	Missing fuel tank cap.	Insecure fuel tank. Dripping fuel leak.
Part 13. General		Serious damage or deterioration that is noticeable and may affect the vehicle's safe operation.
Part 14. Glass and Mirrors	Required mirror or window glass fails to provide the required view to the driver as a result of being cracked, broken, damaged, missing or maladjusted. Required mirror or glass has broken or damaged attachments onto vehicle body.	

Part 15.	. C	D-f
	Control or system failure.	Defroster fails to provide
Heater/Defroster		unobstructed view through the
		windshield.
Part 16. Horn	Vehicle has no operative horn.	
Part 17. Hydraulic	Brake fluid level is below	 Brake boost or power assist
Brake System	indicated minimum level.	is not operative.
-	 Parking brake is inoperative. 	Brake fluid leak.
		Brake pedal fade or
		insufficient brake pedal
		reserve.
		• Activated (other than ABS)
		warning device.
		Brake fluid reservoir is less
D + 10 I	P : 11 1	than ¼ full.
Part 18. Lamps and	• Required lamp does not	When lamps are required:
Reflectors	function as intended.	• Failure of both low-beam
	Required reflector is missing	headlamps.
	or partially missing.	 Failure of both rearmost tail
		lamps.
		At all times:
		Failure of a rearmost turn-
		indicator lamp.
		Failure of both rearmost
		brake lamps.
Part 19. Steering	• Steering wheel lash (free-play)	Steering wheel is insecure,
ruit 19. Steering	is greater than normal.	or does not respond normally.
	is greater than norman.	Steering wheel lash (free-
		play) exceeds required limit.
Dart 20 Sugnancian	. Air look in air ayananaian	Damaged, deflated air bag.
Part 20. Suspension	Air leak in air suspension	C /
System	system.	Cracked or broken main
	Broken spring leaf.	spring leaf or more than one
	Suspension fastener is loose,	broken spring leaf.
	missing or broken.	Part of spring leaf or
		suspension is missing, shifted
		out of place or is in contact
		with another vehicle
		component.
		Loose U-bolt.
Part 21. Tires	Damaged tread or sidewall of	Flat tire.
	tire.	Tire tread depth is less than
	Tire leaking.	wear limit.
		Tire is in contact with
		another tire or any vehicle
		component other than mud-
		flap.
		• Tire is marked "Not for
		highway use".
		Tire has exposed cords in the tread or outer sidewall
Dort 22 W/I I-	. Hyb oil belowii 1.	area.
Part 22. Wheels,	• Hub oil below minimum level	• Wheel has loose, missing or
Hubs and Fasteners	(when fitted with sight glass).	ineffective fastener.
	Leaking wheel seal.	Damaged, cracked or broken
		wheel, rim or attaching part.
		Evidence of imminent
		wheel, hub or bearing failure.
Part 23. Windshield	Control or system malfunction.	When necessary for
Wiper/Washer	Wiper blade damaged, missing	prevailing weather condition.
•	or fails to adequately clear	• Wiper or washer fails to
	driver's field of vision.	adequately clear driver's field
		of vision in area swept by
		driver's side wiper.
<u> </u>	1	

$Schedule \ II-Daily \ Inspection \ of \ Buses$

Application: This Schedule applies to buses designed, constructed and used for the transportation of passengers with a designated seating capacity of more than 10, including the driver, but excluding its operation for personal use. This Schedule also applies to any trailer towed by a bus.

Column 1	Column 2	Column 3
Systems and	Minor Defect(s)	Major Defect(s)
Components	A 71.71% 1	X 1: 1 C:1 4 4
Part 1. Accessibility Devices	Accessibility device may not be used if:	Vehicle fails to return to normal level after "kneeling".
Devices	• Alarm fails to operate.	• Extendable lift, ramp or
	• Equipment malfunctions.	other passenger-loading
	Interlock system malfunctions.	device fails to retract.
Part 2. Air Brake	Audible air leak.	Pushrod stroke of any brake
System	Slow air pressure build-up	exceeds the adjustment limit.
System	rate.	Air loss rate exceeds
	Tuto.	prescribed limit.
		• Inoperative towing vehicle
		(tractor) protection system.
		• Low air warning system
		fails or system is activated.
		Inoperative service, parking
		or emergency brake.
Part 3. Cargo	Insecure or improper load	Insecure cargo.
Securement	covering.	Absence, failure,
		malfunction or deterioration
		of required cargo securement
		device or load covering.
Part 4. Coupling	Coupler or mounting has loose	Coupler is insecure or
Devices	or missing fastener.	movement exceeds prescribed
		limit.
		Coupling or locking
		mechanism is damaged or
		fails to lock.
		Defective, incorrect or
		missing safety chain or cable.
Part 5. Dangerous		Dangerous goods
Goods	D : 1 1 (167)	requirements not met.
Part 6. Doors and	• Door, window or hatch fails to	(Passengers may not be
Emergency Exits	open or close securely.	carried.)
	Alarm inoperative.	Required emergency exit fails to function as intended.
Part 7. Driver	. A coolerator model, alutah	fails to function as intended.
Controls	Accelerator pedal, clutch, gauges, audible and visual	(Passengers may not be carried.)
Controls	indicators or instruments fail to	Accelerator sticking and
	function properly.	engine fails to return to idle.
Part 8. Driver Seat	Seat is damaged or fails to	• Seatbelt or tether belt is
Tuit o. Dirver Seut	remain in set position.	insecure, missing or
	Temam in set position.	malfunctions.
Part 9. Electric Brake	Loose or insecure wiring or	Inoperative breakaway
System	electrical connection.	device.
,		Inoperative brake.
Part 10. Emergency	Emergency equipment is	•
Equipment and	missing, damaged or defective.	
Safety Devices		
Part 11. Exhaust	Exhaust leak.	• Leak that causes exhaust gas
System		to enter the occupant
		compartment.
Part 12. Exterior	Insecure or missing body parts.	Visibly shifted, cracked,
Body and Frame	Insecure or missing	collapsing or sagging frame
	compartment door.	member(s).
	Damaged frame or body.	
Part 13. Fuel System		Missing fuel tank cap.
		• Insecure fuel tank.
		Dripping fuel leak.
Part 14. General		Serious damage or
		deterioration that is noticeable
	Ì	and may affect the vehicle's
D +15 Cl	D : 1 :	safe operation.
Part 15. Glass and	Required mirror or window	safe operation. (Passengers may not be
Part 15. Glass and Mirrors	glass fails to provide the	safe operation. (Passengers may not be carried.)
	glass fails to provide the required view to the driver as a	safe operation. (Passengers may not be carried.) • Driver's view of the road is
	glass fails to provide the required view to the driver as a result of being cracked, broken,	safe operation. (Passengers may not be carried.) • Driver's view of the road is obstructed in the area swept
	glass fails to provide the required view to the driver as a result of being cracked, broken, damaged, missing or	safe operation. (Passengers may not be carried.) • Driver's view of the road is
	glass fails to provide the required view to the driver as a result of being cracked, broken, damaged, missing or maladjusted.	safe operation. (Passengers may not be carried.) • Driver's view of the road is obstructed in the area swept
	glass fails to provide the required view to the driver as a result of being cracked, broken, damaged, missing or maladjusted. • Required mirror or glass has	safe operation. (Passengers may not be carried.) • Driver's view of the road is obstructed in the area swept
	glass fails to provide the required view to the driver as a result of being cracked, broken, damaged, missing or maladjusted. • Required mirror or glass has broken or damaged attachments	safe operation. (Passengers may not be carried.) • Driver's view of the road is obstructed in the area swept
Mirrors	glass fails to provide the required view to the driver as a result of being cracked, broken, damaged, missing or maladjusted. • Required mirror or glass has broken or damaged attachments onto vehicle body.	safe operation. (Passengers may not be carried.) • Driver's view of the road is obstructed in the area swept by the windshield wipers.
Mirrors Part 16.	glass fails to provide the required view to the driver as a result of being cracked, broken, damaged, missing or maladjusted. • Required mirror or glass has broken or damaged attachments	safe operation. (Passengers may not be carried.) • Driver's view of the road is obstructed in the area swept by the windshield wipers. • Defroster fails to provide
Mirrors	glass fails to provide the required view to the driver as a result of being cracked, broken, damaged, missing or maladjusted. • Required mirror or glass has broken or damaged attachments onto vehicle body.	safe operation. (Passengers may not be carried.) • Driver's view of the road is obstructed in the area swept by the windshield wipers.

Part 18. Hydraulic Brake System	Brake fluid level is below indicated minimum level. Parking brake is inoperative.	Brake boost or power assist is not operative. Brake fluid leak. Brake pedal fade or insufficient brake pedal reserve. Activated (other than ABS) warning device. Brake fluid reservoir is less than ¼ full.
Part 19. Lamps and Reflectors	Required interior lamp does not function as intended. Required reflector is missing or partially missing. Passenger safety or access lamp does not function.	When lamps are required: • Failure of both low-beam headlamps. • Failure of both rearmost tail lamps. At all times: • Failure of a rearmost turnindicator lamp. • Failure of both rearmost brake lamps.
Part 20. Passenger Compartment	Stanchion padding is damaged. Damaged steps or floor. Insecure or damaged overhead luggage rack or compartment. Malfunction or absence of required passenger or mobility device restraints. Passenger seat is insecure.	When affected position is occupied: • Malfunction or absence of required passenger or mobility device restraints. • Passenger seat is insecure.
Part 21. Steering	Steering wheel lash (free-play) is greater than normal.	Steering wheel is insecure, or does not respond normally. Steering wheel lash (free-play) exceeds required limit.
Part 22. Suspension System	Air leak in air suspension system. Broken spring leaf. Suspension fastener is loose, missing or broken.	Damaged or deflated air bag. Cracked or broken main spring leaf or more than one broken spring leaf. Part of spring leaf or suspension is missing, shifted out of place or in contact with another vehicle component. Loose U-bolt.
Part 23. Tires	Damaged tread or sidewall of tire. Tire leaking.	Flat tire. Tire tread depth is less than wear limit. Tire is in contact with another tire or any vehicle component other than mudflap. Tire is marked "Not for highway use". Tire has exposed cords in the tread or outer sidewall area.
Part 24. Wheels, Hubs and Fasteners	Hub oil below minimum level (when fitted with sight glass). Leaking wheel seal.	Wheel has loose, missing or ineffective fastener. Damaged, cracked or broken wheel, rim or attaching part. Evidence of imminent wheel, hub or bearing failure.
Part 25. Windshield Wiper/Washer	Control or system malfunction. Wiper blade damaged, missing or fails to adequately clear driver's field of vision.	When necessary for prevailing weather condition. • Wiper or washer fails to adequately clear driver's field of vision in area swept by driver's side wiper.

Schedule III – Daily Inspection of Motor Coaches

Application:
This Schedule applies only to a Motor Coach equipped with air ride suspension, air brakes and automatic brake adjusters. The driver shall be in possession of a valid Schedule IV inspection report or equivalent report(s) in accordance with an approved jurisdictional program. Any trailer towed by a Motor Coach must be inspected in accordance with Schedule II.

Column 1	Column 2	Column 3
Systems and components	Minor Defect(s)	Major Defect(s)
Part 1. Accessibility Devices	Accessibility device may not be used if: • Alarm fails to operate. • Equipment malfunctions. • Interlock system malfunctions.	Vehicle fails to return to normal level after "kneeling". Extendable lift, ramp or other passenger-loading device fails to retract.
Part 2. Air Brake System	Audible air leak. Slow air pressure build-up rate.	There is any indication of a brake adjustment problem. Air loss rate exceeds prescribed limit. Inoperative towing vehicle (tractor) protection system. Low air warning system fails or system is activated. Inoperative service, parking or emergency brake.
Part 3. Coupling Devices	Coupler or mounting has loose or missing fastener.	Coupler is insecure or movement exceeds prescribed limit. Coupling or locking mechanism is damaged or fails to lock. Defective, incorrect or missing safety chain or cable.
Part 4. Dangerous Goods		Dangerous goods requirements not met.
Part 5. Doors and Emergency Exits	Door, window or hatch fails to open or close securely. Alarm inoperative.	(Passengers may not be carried.) • Required emergency exit fails to function as intended.
Part 6. Driver Controls	Accelerator pedal, clutch, gauges, audible and visual indicators or instruments fail to function properly.	(Passengers may not be carried.) • Accelerator sticking and engine fails to return to idle.
Part 7. Driver Seat	Seat is damaged or fails to remain in set position.	Seatbelt or tether belt is insecure, missing or malfunctions.
Part 8. Emergency Equipment and Safety Devices	• Emergency equipment is missing, damaged or defective.	
Part 9. Exhaust System	• Exhaust leak.	Leak that causes exhaust gas to enter the occupant compartment.
Part 10. Exterior Body	Insecure or missing body parts.Insecure or missing compartment door.	
Part 11. Fuel System		 Missing fuel tank cap. Insecure fuel tank. Dripping fuel leak.
Part 12. General		Serious damage or deterioration that is noticeable and may affect the vehicle's safe operation.
Part 13. Glass and Mirrors	Required mirror or window glass fails to provide the required view to the driver as a result of being cracked, broken, damaged, missing or maladjusted. Required mirror or glass has broken or damaged attachments onto vehicle body.	(Passengers may not be carried.) • Driver's view of the road is obstructed in the area swept by the windshield wipers.
Part 14. Heater/Defroster	Control or system failure.	Defroster fails to provide unobstructed view through the windshield.
Part 15. Horn	Vehicle has no operative horn.	
Part 16. Lamps and Reflectors	Required interior lamp does not function as intended. Required reflector is missing or partially missing. Passenger safety or access lamp does not function.	When lamps are required: Failure of both low-beam headlamps. Failure of both rearmost tail lamps. At all times: Failure of a rearmost turnindicator lamp. Failure of both rearmost brake lamps.

Part 17. Passenger Compartment	Stanchion padding is damaged Damaged steps or floor. Insecure or damaged overhead luggage rack or compartment. Malfunction or absence of required passenger or mobility device restraints. Passenger seat is insecure.	When affected position is occupied: • Malfunction or absence of required passenger or mobility device restraints. • Passenger seat is insecure.
Part 18. Suspension System	Air leak in air suspension system.	Damaged, deflated air bag.
Part 19. Steering	Steering wheel lash (free-play) is greater than normal.	Steering wheel is insecure, or does not respond normally. Steering wheel lash (freeplay) exceeds required limit.
Part 20. Tires	Damaged tread or sidewall of tire. Tire leaking.	Flat tire. Tire tread depth is less than wear limit. Tire is in contact with another tire or any vehicle component other than mudflap. Tire is marked "Not for highway use". Tire has exposed cords in the tread or outer sidewall.
Part 21. Wheels, Hubs and Fasteners	Hub oil below minimum level. (When fitted with sight glass.) Leaking wheel seal.	Wheel has loose, missing or ineffective fastener. Damaged, cracked or broken wheel, rim or attaching part. Evidence of imminent wheel, hub or bearing failure.
Part 22. Windshield Wiper/Washer	Control or system malfunction. Wiper blade damaged, missing or fails to adequately clear driver's field of vision.	When necessary for prevailing weather condition. • Wiper or washer fails to adequately clear driver's field of vision in area swept by driver's side wiper.

Schedule IV - Motor Coach (30 days or 12,000 km)

Application: This Schedule applies only to a Motor Coach equipped with air ride suspension, air brakes and automatic brake adjusters.

- Note:
 All conditions listed below are major defects and must be repaired before the vehicle is
- driven.
 Schedule IV inspections must be conducted while the vehicle is positioned over a pit or raised in a manner that provides adequate access to all applicable components by a person who holds the appropriate technician certification or qualification.

Column 1	Column 2
Systems and Components	Major Defect(s)
Part 1. Air Brake System	 Audible air leak. Brake pushrod stroke is at or beyond the adjustment limit. Clearance between disc brake pads and rotor exceeds manufacturer's specified limit. Wedge brake shoe movement exceeds manufacturer's specified limit. Excessive discharge of fluids from air reservoir. Air compressor, mounts or attachments damaged or defective. Compressor drive-belt loose or damaged. Air line or fitting damaged or insecure. Air tank defective, damaged or insecure. Air tank drain or moisture ejector device inoperable. Brake chamber, brake linkage or other brake component is defective, damaged or insecure. DD3 brake chamber fails to hold vehicle in place during tug test, when all air reservoirs are drained. Spring brake is broken or malfunctions. Inoperative service, parking or emergency brake.
Part 2. Exhaust system	Exhaust leak. Exhaust system component insecure, damaged or perforated.
Part 3. Frame and Underbody	Any frame member or fastener is damaged, cracked or insecure. Any component mount is damaged or insecure.

Part 4. Fuel system	Fuel leak.
	Insecure fuel tanks, fuel tank mounts or guards.
	Fuel line or fitting damaged or insecure.
Part 5. Steering	Steering linkage is damaged or insecure.
	Power steering fluid is leaking, contaminated or
	low.
	Power steering component damaged or
	insecure.
Part 6. Suspension System	Air leak or malfunction of air suspension
	system or component.
	Damage or deterioration of any suspension
	component including:
	- spring and air bag;
	 axle or frame attaching component;
	 axle supporting or aligning component;
	 suspension or component fastener;
	 shock absorber or attachments.
Part 7. Tires	Tire inflation less than required.
	Tire treads worn to wear limits.
	Damage to tread or sidewall of tire.
	Retread or rebuilt tire is used on front axle.
Part 8. Wheels and Fasteners	Loose, missing, damaged or ineffective wheel
	fastener.
	Damaged wheel or wheel component.

EXPLANATORY NOTES

SECTION 1 corrects minor wording to the definition for "commercial vehicle" and adds new definitions to the definition section of the regulations.

SECTION 2 clarifies the types of vehicles to which these regulations do not apply.

SECTION 3 corrects wording so the provision is gender neutral.

SECTION 4 amends the references to the Schedules in sections 4 and 5 of the regulations. The Schedule references are changed from Schedule I and II to Schedules I to IV.

SECTION 5 adds a new provision outlining the commercial vehicles to which Schedules I to IV apply.

SECTION 6 revokes and replaces the provisions dealing with the required information to be included in commercial vehicle inspection reports that are prepared by a person who is conducting an inspection of a commercial vehicle in accordance with:

- (1) Schedule I, II or III of the regulations; or
- (2) Schedule IV of the regulations.

The provision also revokes and replaces the sections in the regulations dealing with commercial vehicle inspection reports. An exemption is provided from the requirements to prepare inspection reports in the case of a driver of a truck, tractor and trailer that is being operated within a radius of 160 kilometres of the location at which the driver reported for work if the driver is to return to that location and be released from work.

SECTION 7 corrects the appropriate references to the sections being referred to in the provision. The provision is also updated to provide for the requirement to complete more than one inspection report.

SECTION 8 corrects wording so the provision is gender neutral.

SECTION 9 revokes and replaces the Schedules to the regulations.

SECTION 10 provides for the commencement of these regulations.

HONORARIA AND EXPENSE POLICY COMMISSIONS, BOARDS AND AGENCIES APPROVED

1. Having under consideration the recommendation of Treasury Board (reference Minute TB#279/08 of 1 October 2008) and pursuant to the following statutory citations, Council fixed the remuneration for persons appointed to the following commissions, boards and agencies:

Advisory Council on the Status of Women

via section 8 of the *Advisory Council on the Status of Women Act* R.S.P.E.I. 1988, Cap. A-6

Chairperson \$159 per diem Member \$108 per diem

Agricultural Insurance Corporation

via subsection 4(7) of the *Agricultural Insurance Act* R.S.P.E.I. 1988, Cap. A-8.2

Chairperson \$198 per diem Member \$141 per diem

Agricultural Insurance Corporation Appeal Board

via subsection 14(5) of the *Agricultural Insurance Act* R.S.P.E.I. 1988, Cap. A-8.2

Chairperson \$159 per diem Member \$108 per diem

Apprenticeship Advisory Board, Provincial

via section 6 of the *Apprenticeship and Trades Qualification Act* R.S.P.E.I. 1988, Cap. A-15

Chairperson \$112 per diem Member \$72 per diem

Electoral Boundaries Commission

via subsection 10(1) of the *Electoral Boundaries Act* R.S.P.E.I. 1988, Cap. E-2.1

Member \$141 per diem

Employment Standards Board

via subsection 4(15) of the *Employment Standards Board* R.S.P.E.I. 1988, Cap. E-6.2

Chairperson \$198 per diem Member \$141 per diem

Farm Practices Review Board

via subsection 3(9) of the *Farm Practices Act* R.S.P.E.I. 1988, Cap. F-4.1

Chairperson \$198 per diem Member \$141 per diem

Grain Elevators Corporation, Prince Edward Island

via subsection 2(5) of the *Grain Elevators Corporation Act* R.S.P.E.I. 1988, Cap. G-5

Chairperson \$198 per diem Member \$141 per diem

Heritage Places Advisory Board

via subsection 3(3) of the *Heritage Places Protection Act* R.S.P.E.I. 1988, Cap. H-3.1

Chairperson \$112 per diem Member \$72 per diem

Housing Authorities (various)

via clause 10(3)(e) of the *Housing Corporation Act* R.S.P.E.I. 1988, Cap. H-11.1

Chairperson \$112 per diem Member \$72 per diem

Human Rights Commission, Prince Edward Island

via subsection 16(4) of the *Human Rights Act* R.S.P.E.I. 1988, Cap. H-12

Chairperson \$198 per diem Member \$141 per diem

Interchangeable Drug List Committee

via subsection 28(3) of the *Pharmacy Act* R.S.P.E.I. 1988, Cap. P-6

Physician member \$150 per hour to a maximum

of \$1000 per diem

Pharmacist member \$50 per hour to a maximum

of \$300 per diem

Island Investment Development Inc.

via subsection 3(4) of the *Island Investment Development Act* R.S.P.E.I. 1988, Cap. I-10.1

Chairperson \$198 per diem Member \$141 per diem

Labour Relations Board

via subsection 3(15) of the *Labour Act* R.S.P.E.I. 1988, Cap. L-1

Chairperson \$7,400 per annum and \$200 per diem Member \$141 per diem EXECUTIVE COUNCIL 26 MARCH 2009

Lending Agency Act

via section 4(7) of the *Lending Agency Act* R.S.P.E.I. 1988, Cap. L-6

Chairperson up to \$4,000 per annum prorated

by number of meetings attended and \$175 per diem

Member up to \$2,000 per annum prorated

by number of meetings attended and \$125 per diem

Liquor Control Commission, Prince Edward Island

via clause 4(c) of the *Liquor Control Act* R.S.P.E.I. 1988, Cap. L-14

Chairperson \$198 per diem Member \$141 per diem

Marketing Council, Prince Edward Island

via subsection 2(10) of the *Natural Products Marketing Act* R.S.P.E.I. 1988, Cap. N-3

Chairperson \$198 per diem Member \$141 per diem

Natural Products Appeals Tribunal

via subsection 18(4) of the *Natural Products Marketing Act* R.S.P.E.I. 1988, Cap. N-3

Chairperson \$159 per diem Member \$108 per diem

Physician Resource Planning Committee

via subsection 2.2(6) of *Health Services Payment Act* R.S.P.E.I. 1988, Cap. H-2

Chair \$100 per hour to a maximum

of \$600 per diem

Physician member \$150 per hour to a maximum

of \$1,000 per diem

Power Engineers Board of Examiners

via subsection 4(6) of the *Power Engineers Act* R.S.P.E.I. 1988, Cap. P-15

Member \$72 per diem

Public Forest Council

via section 5 of the *Public Forest Council Act* R.S.P.E.I. 1988, Cap. P-29.1

Chairperson \$112 per diem Member \$72 per diem

Public Service Commission

via subsection 3(6) of the *Civil Service Act* R.S.P.E.I. 1988, Cap. C-8

Chairperson \$112 per diem Members \$72 per diem

Workers Compensation Board of Prince Edward Island

via section 23 of the *Workers Compensation Act* R.S.P.E.I. 1988, Cap. W-7.1

Chairperson \$13,875 per annum

and \$200 per diem

Vice-Chairperson; member \$141 per diem

Workers Compensation Appeal Tribunal

via subsection 56(9) of the *Workers Compensation Act* R.S.P.E.I. 1988, Cap. W-7.1

Chairperson \$7,400 per annum and \$200 per diem

Vice-Chairperson; member \$141 per diem

1. An employee of the province of Prince Edward Island who is appointed to a commission, board or agency, and who is representing the Government of Prince Edward Island, shall not be paid an honorarium under this policy.

2. All rates herein approved come into force effective 1 April 2009.

EC2009-147

PRINCE EDWARD ISLAND LANDS PROTECTION ACT PETITION TO ACQUIRE A LAND HOLDING SUSAN FRANCES MCLEOD (APPROVAL)

Pursuant to section 4 and section 9 of the *Prince Edward Island Lands Protection Act* R.S.P.E.I. 1988, Cap. L-5 Council granted permission to Susan Frances McLeod of Exeter, Rhode Island to acquire a land holding of approximately thirty-two decimal eight (32.8) acres of land in Lot 10, Prince County, Province of Prince Edward Island, being acquired from Patricia Ballem Roy of Ellerslie, Prince Edward Island PROVIDED THAT the said real property is identified for non-development use pursuant to the Land Identification Regulations (EC606/95) made under the said Act.

EC2009-148

PRINCE EDWARD ISLAND LANDS PROTECTION ACT PETITION TO ACQUIRE A LAND HOLDING 101144 P.E.I. INC. (APPROVAL)

Pursuant to section 5 of the *Prince Edward Island Lands Protection Act* R.S.P.E.I. 1988, Cap. L-5 Council granted permission to 101144 P.E.I. Inc. of Flat River, Prince Edward Island to acquire a land holding of approximately zero decimal two six (0.26) acres of land in Lot 48, Queens County, Province of Prince Edward Island, being acquired from Pan American Properties Inc. of Charlottetown, Prince Edward Island.

PRINCE EDWARD ISLAND LANDS PROTECTION ACT PETITION TO ACQUIRE A LAND HOLDING ATKINS ACRES INC. (APPROVAL)

Pursuant to section 5 and section 9 of the *Prince Edward Island Lands Protection Act* R.S.P.E.I. 1988, Cap. L-5 Council granted permission to Atkins Acres Inc. of Glenroy, Prince Edward Island to acquire a land holding of approximately forty-eight decimal four (48.4) acres of land in Lot 38, Kings County, Province of Prince Edward Island, being acquired from Thomas L. Atkins and Patricia A. Atkins, both of Glenroy, Prince Edward Island PROVIDED THAT the said real property is identified for non-development use pursuant to the Land Identification Regulations (EC606/95) made under the said Act.

EC2009-150

PRINCE EDWARD ISLAND LANDS PROTECTION ACT PETITION TO ACQUIRE A LAND HOLDING CRAGGAN FARMS LTD. (APPROVAL)

Pursuant to section 5 and section 9 of the *Prince Edward Island Lands Protection Act* R.S.P.E.I. 1988, Cap. L-5 Council granted permission to Craggan Farms Ltd. of Marshfield, Prince Edward Island to acquire a land holding of approximately eighteen (18) acres of land in Lot 34, Queens County, Province of Prince Edward Island, being acquired from David Wood and John Wood, both of Marshfield, Prince Edward Island PROVIDED THAT the said real property is identified for non-development use pursuant to the Land Identification Regulations (EC606/95) made under the said Act.

EC2009-151

PRINCE EDWARD ISLAND
LANDS PROTECTION ACT
PETITION TO ACQUIRE A LAND HOLDING
ICE IMPORTS LTD.
(APPROVAL)

Pursuant to section 5 of the *Prince Edward Island Lands Protection Act* R.S.P.E.I. 1988, Cap. L-5 Council granted permission to ICE Imports Ltd. of Charlottetown, Prince Edward Island to acquire a land holding of approximately zero decimal six five (0.65) acres of land in Charlottetown, Queens County, Province of Prince Edward Island, being acquired from 100209 P.E.I. Inc. of Charlottetown, Prince Edward Island.

PRINCE EDWARD ISLAND LANDS PROTECTION ACT PETITION TO ACQUIRE A LAND HOLDING LES INVESTISSEMENTS NORMAND BEAUREGARD INC. (APPROVAL)

Pursuant to section 5 of the Prince Edward Island Lands Protection Act R.S.P.E.I. 1988, Cap. L-5 Council granted permission to Les Investissement Normand Beauregard Inc. of Sutton, Quebec to acquire a land holding of approximately seven decimal two three (7.23) acres of land in Lot 16, Prince County, Province of Prince Edward Island, being acquired from Les Investissements Normand Beauregard Inc., Mortgagee in Possession, of Sutton, Quebec.

EC2009-153

PRINCE EDWARD ISLAND LANDS PROTECTION ACT PETITION TO ACQUIRE A LAND HOLDING LA SOCIÉTÉ DE DÉVELOPPEMENT DE LA BAIE ACADIENNE INCORPORÉE (APPROVAL)

Pursuant to section 5 of the Prince Edward Island Lands Protection Act R.S.P.E.I. 1988, Cap. L-5 Council granted permission to La Société de développement de la Baie acadienne incorporée of Wellington, Prince Edward Island to acquire a land holding of approximately seven decimal two three (7.23) acres of land in Lot 16, Prince County, Province of Prince Edward Island, being acquired from Les Investissements Normand Beauregard Inc., Mortgagee in Possession, of Sutton, Quebec.

EC2009-154

PRINCE EDWARD ISLAND LANDS PROTECTION ACT PETITION TO ACQUIRE A LAND HOLDING LONG RIVER FARMS LTD. (APPROVAL)

Pursuant to section 5 of the Prince Edward Island Lands Protection Act R.S.P.E.I. 1988, Cap. L-5 Council granted permission to Long River Farms Ltd. of Charlottetown, Prince Edward Island to acquire a land holding of approximately one hundred and four (104) acres of land in Lot 18, Prince County, Province of Prince Edward Island, being acquired from Riley Farms Ltd. of Kensington, Prince Edward Island.

Further, Council noted that the said land holding, being Provincial Property Nos. 83832 and 84244, was previously identified for non-development use in accordance with section 2 of the Land Identification Regulations (EC606/95) made under the said Act. Identification continues to apply.

PRINCE EDWARD ISLAND LANDS PROTECTION ACT PETITION TO ACQUIRE A LAND HOLDING MACLENNAN INVESTMENTS LTD. (APPROVAL)

Pursuant to section 5 of the *Prince Edward Island Lands Protection Act* R.S.P.E.I. 1988, Cap. L-5 Council granted permission to MacLennan Investments Ltd. of Charlottetown, Prince Edward Island to acquire a land holding of approximately zero decimal six five (0.65) acres of land in Charlottetown, Queens County, Province of Prince Edward Island, being acquired from Two Three Four Five Six P.E.I. Ltd. of Charlottetown, Prince Edward Island.

EC2009-156

PRINCE EDWARD ISLAND
LANDS PROTECTION ACT
PETITION TO ACQUIRE A LAND HOLDING
MARTIN RD FARMS LTD.
(APPROVAL)

Pursuant to section 5 and section 9 of the *Prince Edward Island Lands Protection Act* R.S.P.E.I. 1988, Cap. L-5 Council granted permission to Martin Rd Farms Ltd. of St. Roch, Prince Edward Island to acquire a land holding of approximately twenty-one decimal six four (21.64) acres of land in Lot 2, Prince County, Province of Prince Edward Island, being acquired from LeRoy Martin and Christine Martin, both of St. Roch, Prince Edward Island PROVIDED THAT the said real property is identified for non-development use pursuant to the Land Identification Regulations (EC606/95) made under the said Act.

EC2009-157

PRINCE EDWARD ISLAND LANDS PROTECTION ACT PETITION TO ACQUIRE A LAND HOLDING MARTIN RD FARMS LTD. (APPROVAL)

Pursuant to section 5 and section 9 of the *Prince Edward Island Lands Protection Act* R.S.P.E.I. 1988, Cap. L-5 Council granted permission to Martin Rd Farms Ltd. of St. Roch, Prince Edward Island to acquire a land holding of approximately one hundred and eighty-eight decimal four four (188.44) acres of land in Lot 2, Prince County, Province of Prince Edward Island, being acquired from LeRoy Martin of St. Roch, Prince Edward Island PROVIDED THAT the said real property is identified for non-development use pursuant to the Land Identification Regulations (EC606/95) made under the said Act.

PRINCE EDWARD ISLAND LANDS PROTECTION ACT PETITION TO ACQUIRE A LAND HOLDING MILTONVALE PARK COMMUNITY COUNCIL (APPROVAL)

Pursuant to section 5 of the *Prince Edward Island Lands Protection Act* R.S.P.E.I. 1988, Cap. L-5 Council granted permission to Miltonvale Park Community Council of Winsloe, Prince Edward Island to acquire, by lease, an interest in a land holding of approximately ninety-seven decimal two seven (97.27) acres of land in Lot 32, Queens County, Province of Prince Edward Island, being acquired from the Government of Prince Edward Island, as represented by the Minister of Transportation and Public Works of Charlottetown, Prince Edward Island.

EC2009-159

PRINCE EDWARD ISLAND LANDS PROTECTION ACT APPLICATION TO LEASE LAND SOMERGLEN FARMS LTD. (TO RESCIND)

Council, having under consideration Order-in-Council EC2006-303 of 23 May 2006, rescinded the said Order forthwith, thus rescinding permission for Somerglen Farms Ltd. to acquire, by lease, an interest in up to two hundred and fifty (250) acres as part of its aggregate land holdings.

EC2009-160

PRINCE EDWARD ISLAND LANDS PROTECTION ACT APPLICATION TO LEASE LAND SOMERGLEN FARMS LTD. (APPROVAL)

Pursuant to section 5 and clause 5.3(1)(b) of the *Prince Edward Island Lands Protection Act* R.S.P.E.I. 1988, Cap. L-5 Council granted permission to Somerglen Farms Ltd. of St. Georges, Prince Edward Island to acquire, by lease, an interest in a land holding or land holdings of up to three hundred (300) acres of land as part of the said corporation's aggregate land holdings PROVIDED THAT the said Somerglen Farms Ltd. files a statement with the Island Regulatory and Appeals Commission within one year of the date of this Order and prior to 31 December in every subsequent year disclosing the parcel number, the acreage and the term of lease for each parcel leased during the reporting period covered by the statement.

PRINCE EDWARD ISLAND LANDS PROTECTION ACT PROPERTY NO. 082636, LOT 25, PRINCE COUNTY IDENTIFICATION FOR NON-DEVELOPMENT USE AMENDMENT

Pursuant to subsection 9(2) of the *Prince Edward Island Lands Protection Act* R.S.P.E.I. 1988, Cap. L-5, Council amended the condition of non-development use made pursuant to section 2 of the Land Identification Regulations (EC606/95) in respect of approximately one hundred and sixty four decimal five (164.5) acres of land, being Provincial Property No. 082636 located in Lot 25, Prince County, Prince Edward Island and currently owned by Willard Waugh & Sons Ltd. of Summerside, Prince Edward Island.

Council noted that this amendment will enable subdivision of a parcel of land of approximately zero decimal one eight (0.18) acres, SUBJECT TO the subdivided parcel being appended to the adjacent Provincial Property No. 590612. Further, Council determined that following subdivision, identification for non-development use shall continue to apply to the remaining land.

This Order-in-Council comes into force on 26 March 2009.

EC2009-162

OIL AND NATURAL GAS ACT PERMIT, LEASE AND SURVEY SYSTEM REGULATIONS AMENDMENT

Pursuant to section 11 of the *Oil and Natural Gas Act* R.S.P.E.I. 1988, Cap. O-5, Council made the following regulations:

- 1. Section 8 of the *Oil and Natural Gas Act* Permit, Lease and Survey System Regulations (EC689/00) is amended
 - (a) by renumbering it as subsection 8(1); and
 - (b) by the addition of the following after subsection (1):
- (2) Notwithstanding subsection (1), the Minister may grant a one-time One-time extension extension to the term of an onshore permit prescribed by subsection (1), for a term not to exceed two years from the date of expiry of the permit, with such conditions as the Minister may impose.
- 2. These regulations come into force on April 4, 2009.

EXPLANATORY NOTES

SECTION 1 authorizes the Minister to extend the term of an onshore permit for a further term not to exceed two years, with such conditions as the Minister may impose.

SECTION 2 provides for the commencement of these regulations.

POLICE ACT PROVINCIAL POLICE CONSTABLE APPOINTMENT

Pursuant to section 6 of the *Police Act* R.S.P.E.I. 1988, Cap. P-11 Council authorized the following person to exercise the powers of a provincial police constable:

NAME

TERM OF APPOINTMENT

Brian J. Donnelly

from 26 March 2009 at pleasure

Further, Council ordered that, should the aforementioned person cease to be employed with the City of Charlottetown Police Services, his appointment as a provincial police constable shall terminate coincident with the date his employment with the City of Charlottetown Police Services is terminated.

EC2009-164

PROVINCIAL COURT ACT JUSTICE OF THE PEACE APPOINTMENT

Under authority of section 14 of the *Provincial Court Act* R.S.P.E.I. 1988, Cap. P-25 Council appointed Karen Jennifer McInnis of Summerside in Prince County, Prince Edward Island, as a Justice of the Peace in and for the Counties of Prince, Queens and Kings in the Province of Prince Edward Island for a term of five (5) years, effective 26 March 2009.

Further, Council ordered that should Karen Jennifer McInnis cease to be employed in her present capacity with the Office of the Attorney General, that her appointment as a Justice of the Peace shall terminate coincident with the date her employment terminates.

EC2009-165

PROVINCIAL COURT ACT JUSTICE OF THE PEACE APPOINTMENT

Under authority of section 14 of the *Provincial Court Act* R.S.P.E.I. 1988, Cap. P-25 Council appointed Tammy Vessey of Charlottetown, in Queens County, Prince Edward Island, as a Justice of the Peace for the City of Charlottetown, in the Province of Prince Edward Island, for a term of five (5) years, effective 26 March 2009.

Further, Council ordered that the appointment of Tammy Vessey be limited to matters related to enforcement of the City of Charlottetown bylaws, and should she cease to be employed in her present capacity with the City of Charlottetown, that her appointment as a Justice of the Peace shall terminate coincident with the date her employment terminates.

PROVINCIAL COURT ACT AND VICTIMS OF FAMILY VIOLENCE ACT JUSTICES OF THE PEACE APPOINTMENTS AND DESIGNATIONS

Under authority of section 14 of the *Provincial Court Act* R.S.P.E.I. 1988, Cap. P-25, Council appointed the following persons as Justices of the Peace, each for a term of three years, effective 26 March 2009, and designated them pursuant to section 14 of the Victims of Family Violence Act R.S.P.E.I. 1988, Cap. V-3.2, as justices of the peace to hear and determine applications within the Province of Prince Edward Island:

> R. Lee Forbes, Charlottetown Robert H. Gibson, Stratford Emil Anton Halvorson, Charlottetown Hank Spierenburg, Charlottetown

EC2009-167

ROADS ACT HIGHWAY ACCESS REGULATIONS **AMENDMENT**

Pursuant to subsection 29(1) of the Roads Act R.S.P.E.I. 1988, Cap. R-15, Council made the following regulations:

- 1. Subsection 20(1) of the Roads Act Highway Access Regulations (EC580/95) is amended by the addition of the following after clause (c):
 - (c.1) to establish an institutional operation on a parcel of land where there is no existing institutional operation and it has not been determined by the Lieutenant Governor in Council that establishment of the institutional operation along an arterial highway that lies west of the intersection of Route 2 and Route 124 in Prince County, or east of the intersections of Route 1 and Route 3 or Route 2 and Route 6 in Queens County, is in the best interest of the
- 2. These regulations come into force on April 4, 2009.

EXPLANATORY NOTES

SECTION 1 adds a provision to authorize the establishment of an institutional operation on a parcel of land where there is no existing institutional operation.

SECTION 2 provides for the commencement of these regulations.

SUMMARY PROCEEDINGS ACT TICKET REGULATIONS **AMENDMENT**

Pursuant to subsection 10(3) of the Summary Proceedings Act R.S.P.E.I. 1988, Cap. S-9, Council made the following regulations:

1. Schedule 2 of the Summary Proceedings Act Ticket Regulations (EC58/08) is amended, in Part 30, by the revocation of items 1 to 7 and the substitution of the following:

1	Driver failing to inspect vehicle or cause it to be inspected on a daily basis prior to first trip of day	3(1)	200
2	Driver failing to inspect vehicle on continuous trip later than the first rest stop of the day		200
3	Driver or person designated by driver failing to enter defects	3(2)	
4	in inspection report Person failing to ensure inspection includes every system and	5	200
5	component listed in inspection schedule Person failing to ensure inspection adequate to determine whether there is a minor defect or major defect as set out	5.1(2)	200
6	in inspection schedule	5.1(3)	200
7	every system and component listed in Schedule IV Person failing to ensure inspection adequate to determine whether there is a major defect as set out in Schedule	5.1(6)	200
8	IV Person failing to ensure defect detected in under-vehicle inspection is repaired before motor coach is driven or	5.1(7)	200
9	operated	5.1(8)	500
	while in charge of vehicle Person conducting inspection failing to prepare an inspection	6	200
	report in accordance with Schedule I, II or III, as required	7(1)	200
11	Person conducting under-vehicle inspection failing to prepare an inspection report in accordance with Schedule IV, as		
12	required	7(2)	200
13	where vehicle not inspected	7(3)	500
13	where person not in possession of required inspection	5 (1)	200
14	reports and Schedules	7(4)	200
15	an inspector	7(5)	200
	detected during inspection	8(1)	200
16	Person conducting inspection failing to record defects detected during inspection and failing to report defects to		
17	carrier	8(2)	200
10	repairs carried out	8(3)	200
18	Driver failing to record defects detected by driver on inspection report and failing to report defects to carrier		
19	prior to next required inspection	8(4)	200
	in operation on inspection report and failing to immediately report major defects to carrier	8(5)	200
20	Carrier permitting person to drive and person driving vehicle where major defect present on the vehicle	8(6)	500
21	Carrier failing to ensure previously reported defects rectified prior to next required inspection or within timeframe	8(7)	200
22	Driver failing to provide original copy of inspection report to carrier within 20 days after date inspection report is	` , ,	
23	completed	9(1)	200
	certification of repairs for at least six months	9(2)	200
∠ ¬	inspection reports at carrier's principal place of business	1.1	200
	for required time-frames	11	200

2. These regulations come into force on April 4, 2009.

EXPLANATORY NOTES

The amendment updates the ticketable offences outlined in the Commercial Vehicle (Trip Inspection and Records) Regulations.

WILDLIFE CONSERVATION ACT ANGLING REGULATIONS **AMENDMENT**

Pursuant to section 28 of the Wildlife Conservation Act R.S.P.E.I. 1988, Cap. W-4.1, Council made the following regulations:

- 1. Section 1 of the Wildlife Conservation Act Angling Regulations (EC180/02) is amended
 - (a) by the revocation of clause (b);
 - (b) by renumbering clause (c) as clause (b);
 - (c) by the revocation of clauses (d) to (g);
 - (d) by renumbering clause (h) as clause (c);
 - (e) by the addition of the following after clause (c):
 - (d) "perch" means the species of perch known as white perch having perch the scientific name Morone americana;
 - (f) by renumbering clause (i) as clause (e);
 - (g) by the revocation of clause (j); and
 - (h) by renumbering clause (k) as clause (f).
- 2. (1) Subsection 2(1) of the regulations is amended
 - (a) by the deletion of the words "trout angling licenses and salmon angling licenses" and the substitution of the words "angling licenses for trout, perch and salmon"; and
 - (b) by renumbering it as section 2.
 - (2) Subsection 2(2) of the regulations is revoked.
- 3. Sections 3 to 13 of the regulations are revoked and the following substituted:

ICE FISHING

- 3. No person shall, when angling for trout or perch through ice,
- Angling through ice

Prohibition

- (a) leave any shelter materials unattended;
- (b) use shacks for the purpose of angling; or
- (c) leave holes cut in the ice unless marked with a spruce bough or some other appropriate natural vegetation.

GENERAL PROHIBITIONS

- **4.** No person shall
 - (a) damage or obstruct
 - (i) any fish-way,
 - (ii) any device used to monitor fish passage, or
 - (iii) any device used by fish to pass over or around any obstruction;
 - (b) do anything to stop or hinder fish
 - (i) from entering or passing any fish-way, or
 - (ii) from surmounting any obstacle or leap;
 - (c) angle for or attempt to catch fish in any manner in any fish-way;
 - (d) catch or attempt to catch trout or salmon by means of chumming;
 - (e) while angling, be in possession of or retain trout unless it is immediately killed or released into the water;
 - (f) sell, possess or use fish eggs as bait for angling; or
 - (g) litter while angling.
- **5.** These regulations do not apply to licensed private fishing preserves.

Exception

- 4. The Schedule to the regulations is revoked and the Schedule as set out in the Schedule to these regulations is substituted.
- 5. These regulations come into force on April 4, 2009.

Fee-GST

SCHEDULE

SCHEDULE LICENSE FEES

		LICENSE	Included		
1.	ANGLIN	NG FOR TROUT AND PERCH			
	(i)	Resident or non-resident	\$10.00		
	(ii)	Family (5 days), Available May 1	5.00		
	(iii)	Courtesy resident (16 to 18 years of age and 60 years of age or older)	Free		
2.	SALMON FLY FISHING				
	(i)	Resident	\$10.00		
	(ii)	Non-resident	10.00		
	For the pu endants.	rpose of clause 1(ii), "family" includes the license bearer	's spouse and	family includes	
4. All persons applying for a salmon fly fishing license must also have an angling license or be exempt under the Act.				Salmon license - requirements	

EXPLANATORY NOTES

SECTION 1 revokes definitions in the definition section of the regulations which are no longer required. A definition for "perch" is added to this provision.

SECTION 2 amends the type of license which the Minister may issue. Trout angling licenses and salmon angling licenses have been replaced by an angling license for trout, perch and salmon. A subsection dealing with a salmon angling license is revoked, which has become redundant.

SECTION 3 revokes provisions which have become unnecessary and replaces them with two offence provisions dealing with offences in respect of angling for trout or perch through ice, and a general offence provision. A section is added to clarify that the regulations do not apply to licensed private fishing preserves.

SECTION 4 revokes and replaces the Schedule to the regulations dealing with license fees.

SECTION 5 provides for the commencement of these regulations.

EC2009-170

WILDLIFE CONSERVATION ACT WILDLIFE CONSERVATION FUND REGULATIONS

Pursuant to section 28 of the *Wildlife Conservation Act* R.S.P.E.I. 1988, Cap. W-4.1, Council made the following regulations:

1. (1) The Wildlife Conservation Fund is continued pursuant to the Wildlife Wildlife Conservation Act R.S.P.E.I. 1988, Cap. W-4.1.

Wildlife Conservation Fund

- (2) Every person who obtains a license or a permit for angling, hunting Fees or trapping pursuant to the *Wildlife Conservation Act* or any regulations made or continued under the Act, shall pay a Wildlife Conservation Fund fee
 - (a) in the amount of \$13.00 (GST included) where the person is 65 years of age or older;
 - (b) in the amount of \$20.00 (GST included) where the person is between 16 and 65 years of age; or
 - (c) in the amount of \$5.00 (GST included) for a Family Wildlife Conservation Fund license where the person purchases a family angling license.
- (3) Wildlife Conservation Fund fees collected pursuant to subsection Fees paid to Fund (2) shall be paid into the Wildlife Conservation Fund.
- (4) Where a person has paid the Wildlife Conservation Fund fee Wildlife/Family pursuant to subsection (2), the Department shall issue
 - (a) a Wildlife Conservation Fund license valid for the person license named on the license; or

Wildlife/Family Wildlife Conservation Fund license

- (b) a Family Wildlife Conservation Fund license valid for the person named on the license and the person's spouse and dependents.
- (5) Every Wildlife Conservation Fund license expires on March 31 of Expiry of license the year immediately following the issue of the license.
- (6) Every Family Wildlife Conservation Fund license expires five days *Idem* after the date of the issue of the license.
- (7) No person shall hunt or trap without being in possession of a valid Prohibition Wildlife Conservation Fund license.
- (8) No person shall angle without being in possession of a valid *Idem* Wildlife Conservation Fund license or a valid Family Wildlife Conservation Fund license
- (9) Where a person holds more than one license pursuant to the Act or Multiple licenses any regulations made or continued pursuant to the Act, the person is required to hold only one valid Wildlife Conservation Fund license.
- (10) These regulations do not apply to aboriginal persons or persons Exception under the age of sixteen years.
- **2.** The *Wildlife Conservation Act* Wildlife Conservation Fund Revocation Regulations (EC211/99) are revoked.
- **3.** These regulations come into force on April 4, 2009.

Commencement

EXPLANATORY NOTES

SECTION 1 outlines the Wildlife Conservation Fund fees.

The provision provides that the Department shall issue a Wildlife Conservation Fund license which is valid for the person named on the license and also a Family Wildlife Conservation Fund license which is valid for the person named on the license as well as the person's spouse and dependents.

Provisions provide for the expiry of a Wildlife Conservation Fund license and a Family Wildlife Conservation Fund license.

The offence provisions provide for the prohibition against hunting or trapping without being in possession of a valid Wildlife Conservation Fund license, as well as a prohibition against angling without being in possession of a valid Wildlife Conservation Fund license or a valid Family Wildlife Conservation Fund license.

The provision dealing with multiple licenses provides that a multiple license holder is only required to hold one valid Wildlife Conservation Fund license.

A provision provides that the regulations do not apply to aboriginal persons or persons under the age of sixteen years.

SECTION 2 revokes the *Wildlife Conservation Act* Conservation Fund Regulations (EC211/99).

SECTION 3 provides for the commencement of these regulations.

WILDLIFE CONSERVATION ACT WILDLIFE MANAGEMENT AREAS REGULATIONS

Pursuant to clause 16(1)(c) of the *Wildlife Conservation Act* R.S.P.E.I. 1988, Cap. W-4.1, Council made the following regulations:

1. The lands described in Schedules A and B are designated as wildlife Designation management areas.

SCHEDULE A

ELLENS CREEK - BRIGHTON SHORE WILDLIFE MANAGEMENT AREA

1. The following area of land is designated as a wildlife management area:

DESIGNATION

COMMENCING at the point of intersection of the ordinary high water mark of the North or Yorke River with the extension westerly of the center line of Brighton Road.

THENCE westerly on the line of extension of the center line of Brighton Road 665 yards or until intersected by a line extending southwardly from the southeast corner of Parcel #388900 on Lewis Point, said line being a minimum distance of 450 yards westerly of the most northerly lighthouse on Queen Elizabeth Drive.

THENCE northerly along said line at a minimum distance of 450 yards westerly of the said most northerly lighthouse on Queen Elizabeth Drive until said line intersects the ordinary highwater mark at Lewis Point.

THENCE northerly following the various courses of the ordinary high water mark of the North or Yorke River and Ellens Creek, crossing the Beach Grove Road, to the southern boundary of the Trans Canada Highway.

THENCE continuing to follow the various courses of the ordinary high water mark of the said Ellens Creek and North or Yorke River southerly to the place of commencement.

Being and intended to be all of the area of water enclosed within the above described boundaries and as further delineated, shaded, and crosshatched on the following map reference.

REGULATIONS

- **1.1** No person shall discharge a firearm from or within the boundaries of the Ellens Creek Brighton Shore Wildlife Management Area.
- **1.2** No person shall hunt, or otherwise molest or disturb wildlife in the Ellens Creek Brighton Shore Wildlife Management Area provided that nothing in these regulations shall prohibit any person from angling sports fish or fishing for shellfish, in accordance with the *Fisheries Act* (Canada) and the *Wildlife Conservation Act* Angling Regulations or from trapping furbearing animals, in accordance with the *Wildlife Conservation Act* Fur Harvesting Regulations.
- **1.3** Every person who violates or attempts to violate any provision of these regulations is guilty of an offence and is liable, on summary conviction, to the fines specified under section 32 of the Act.

FOREST HILL WILDLIFE MANAGEMENT AREA

2. The following area of land is designated as a wildlife management area:

DESIGNATION

ALL THOSE PARCELS OF LAND situate, lying, and being on Lot or Township No. 55 in Kings County, Province of Prince Edward Island, described as follows, that is to say:

PARCEL 1:

Includes all lands identified as provincial property number 122556, consisting of approximately 15.4 hectares (38.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 21st day of December, A.D. 1994 from the Prince Edward Island Agricultural Insurance Corporation and the Canada - Prince Edward Island Crop Insurance Fund to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 2439 with the Registrar of Deeds for Kings County on the 21st day of December, A.D. 1994 in Book 298, Page 67.

PARCEL 2:

Includes all lands identified as provincial property number 122564, consisting of approximately 12.1 hectares (30.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 21st day of December, A.D. 1994 from the Prince Edward Island Agricultural Insurance Corporation and the Canada - Prince Edward Island Crop Insurance Fund to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 2439 with the Registrar of Deeds for Kings County on the 21st day of December, A.D. 1994 in Book 298, Page 67.

PARCEL 3:

Includes all lands identified as provincial property number 155291, consisting of approximately 48.6 hectares (120.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 21st day of December, A.D. 1994 from the Prince Edward Island Agricultural Insurance Corporation and the Canada - Prince Edward Island Crop Insurance Fund to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 2439 with the Registrar of Deeds for Kings County on the 21st day of December, A.D. 1994 in Book 298, Page 67.

PARCEL 4:

Includes all lands identified as provincial property number 155465, consisting of approximately 80.9 hectares (200.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 4th day of April, A.D. 1985 from John Sterling MacDonald and Annie K. MacDonald to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 421 with the Registrar of Deeds for Kings County on the 4th day of April, A.D. 1985 in Book 182, Page 08.

PARCEL 5:

Includes all lands identified as provincial property number 155499, consisting of approximately 58.3 hectares (144.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 26th day of April, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 698 with the Registrar of Deeds for Kings County on the 27th day of April, A.D. 1994 in Book 287, Page 89.

PARCEL 6:

Includes all lands identified as provincial property number 156596, consisting of approximately 25.8 hectares (63.7 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 21st day of December, A.D. 1994 from the Prince Edward Island Agricultural Insurance Corporation and the Canada - Prince Edward Island Crop Insurance Fund to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 2439 with the Registrar of Deeds for Kings County on the 21st day of December, A.D. 1994 in Book 298, Page 67.

PARCEL 7:

Includes all lands identified as provincial property number 165241, consisting of approximately 23.5 hectares (58.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 21st day of December, A.D. 1994 from the Prince Edward Island Agricultural Insurance Corporation and the Canada - Prince Edward Island Crop Insurance Fund to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 2439 with the Registrar of Deeds for Kings County on the 21st day of December, A.D. 1994 in Book 298, Page 67.

PARCEL 8:

Includes all lands identified as provincial property number 165258, consisting of approximately 129.5 hectares (320.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 25th day of October, A.D. 1968 from Annie F. Howlett to Her Majesty the Queen and registered as Document Number 985 with the Registrar of Deeds for Kings County on the 22nd day of November, A.D. 1968 in Book 88, Page 355.

PARCEL 9:

Includes all lands identified as provincial property number 411215, consisting of approximately 25.5 hectares (63.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 26th day of April, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 698 with the Registrar of Deeds for Kings County on the 27th day of April, A.D. 1994 in Book 287, Page 89.

PARCEL 10:

Includes all lands identified as provincial property number 436089, consisting of approximately 13.4 hectares (33.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 26th day of May, A.D. 1972 from John J. Reilly and Norman E. Reilly to Her Majesty the Queen in the Right of the Province of Prince Edward Island and registered as Document Number 578 with the Registrar of Deeds for Kings County on the 26th day of May, A.D. 1972 in Book 93, Page 604.

PARCEL 11:

Includes all lands identified as provincial property number 488262, consisting of approximately 80.9 hectares (200.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 15th day of March, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 433 with the Registrar of Deeds for Kings County on the 15th day of March, A.D. 1994 in Book 286, Page 29.

PARCEL 12:

Includes all lands identified as provincial property number 500413, consisting of approximately 12.1 hectares (30.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 26th day of April, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 698 with the Registrar of Deeds for Kings County on the 27th day of April, A.D. 1994 in Book 287, Page 89.

PARCEL 13:

Includes all lands identified as provincial property number 165217, consisting of approximately 17.2 hectares (42.6 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 23rd day of October, A.D. 1995 from Newsco Investments Ltd to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 1963 with the Registrar of Deeds for Kings County on the 23rd day of October, A.D. 1995 in Book 317, Page 54.

PARCEL 14:

Includes all lands identified as provincial property number 165118, consisting of approximately 34.0 hectares (84.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 23rd day of October, A.D. 1995 from Newsco Investments Ltd to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 1963 with the Registrar of Deeds for Kings County on the 23rd day of October, A.D. 1995 in Book 317, Page 54.

PARCEL 15:

Includes all lands identified as provincial property number 799205, consisting of approximately 23.1 hectares (57.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 18th day of December, A.D. 1992 from Garrett Farms Ltd to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 2091 with the Registrar of Deeds for Kings County on the 21st day of December, A.D. 1992 in Book 270, Page 98.

PARCEL 16:

Includes all lands identified as provincial property number 155416, consisting of approximately 11.6 hectares (28.6 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 23rd day of October, A.D. 1995 from Newsco Investments Ltd to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 1963 with the Registrar of Deeds for Kings County on the 23rd day of October, A.D. 1995 in Book 317, Page 54.

PARCEL 17:

Includes all lands identified as provincial property number 155366, consisting of approximately 19.0 hectares (47.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance from Wynne G. Potter to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 564 with the Registrar of Deeds for Kings County on the 29th day of March, A.D. 1996 in Book 327, Page 33.

PARCEL 18:

Includes all lands identified as provincial property number 155382, consisting of approximately 32.4 hectares (80.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 26th day of April, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 698 with the Registrar of Deeds for Kings County on the 27th day of April, A.D. 1994 in Book 287, Page 89.

PARCEL 19:

Includes all lands identified as provincial property number 155408, consisting of approximately 4.0 hectares (10.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 17th day of March, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 449 with the Registrar of Deeds for Kings County on the 18th day of April, A.D. 1994 in Book 286, Page 38.

PARCEL 20:

Includes all lands identified as provincial property number 155374, consisting of approximately 8.1 hectares (20.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 26th day of April, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 698 with the Registrar of Deeds for Kings County on the 27th day of April, A.D. 1994 in Book 287, Page 89.

PARCEL 21:

Includes all lands identified as provincial property number 483784, consisting of approximately 43.7 hectares (108.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 17th day of March, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 449 with the Registrar of Deeds for Kings County on the 18th day of April, A.D. 1994 in Book 286, Page 38.

PARCEL 22:

Includes all lands identified as provincial property number 155333, consisting of approximately 20.2 hectares (50.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 7th day of September, A.D. 1990 from Svend Christensen to Carl Stephen Hans Christensen and registered as Document Number 1548 with the Registrar of Deeds for Kings County on the 11th day of September, A.D. 1990 in Book 244, Page 14.

GLENFINNAN WILDLIFE MANAGEMENT AREA

3. The following area of land is designated as a wildlife management area:

DESIGNATION

ALL THAT TRACT OF LAND situate, lying and being in Glenfinnan on Lot 35 in Queens County in the Province of Prince Edward Island, bounded and described as follows:

COMMENCING at a point on the eastern boundary of the Glenfinnan Road, said point being at a distance of forty-two chains and forty-two links northwardly from the northern side of the old road leading to Glenfinnan Bridge and said point being also at the northwestern corner of a 60-acre tract of land in the possession of the estate of John W. MacDonald;

THENCE running south 80 30' east (according to the magnetic meridian of the year 1764) along a wire fence which marks the northern boundary of the above-mentioned 60-acre tract of land a distance of about 43 chains or to the western shore of Glenfinnan River:

THENCE running northwardly along the various courses of the said western shore of Glenfinnan River and westwardly along the various courses of the southern shore of the Hillsboro River until it meets a point that is in the continuation northwardly of the eastern boundary of the Glenfinnan Road;

AND THENCE running south 7 30' west along the said continuation northwardly and the eastern boundary of the Glenfinnan Road a distance of thirty-seven chains and ninety-five links or to the point at the place of commencement.

AND containing by estimation ninety-two acres of arable land and approximately thirty acres of tidal marsh, a little more or less. (EC1203/66)

REGULATIONS

- **3.1** The open season for the taking and shooting of pheasant on the Glenfinnan Wildlife Management Area shall extend from October 14 to January 3 next following, both dates inclusive.
- **3.2** The fees for a license to hunt pheasant on the Glenfinnan Wildlife Management Area are two dollars for a resident and three dollars for a non-resident.
- **3.3** An additional fee of three dollars per bird released shall be charged with a minimum release of two birds per hunter. Each hunter will be guaranteed a minimum of two birds.
- **3.4** No person shall undertake to hunt on the Glenfinnan Wildlife Management Area unless the person is a holder of either a resident or a non-resident hunting license issued by the Minister under the authority of section 11 of the *Wildlife Conservation Act* and has paid the fees as prescribed above.
- **3.5** No person shall remove any pheasant from the Glenfinnan Wildlife Management Area that does not have attached to the leg a metal seal bearing the area identification number.
- **3.6** Hunters shall be accommodated on a first come, first served basis and reservations will be required at least one day in advance of the desired hunting date.
- **3.7** Each hunter will be required to register with the attending officer at the Glenfinnan Wildlife Management Area before proceeding to hunt and upon payment of the prescribed fees will proceed to hunt in the area allotted to the hunter by the officer. A maximum of three hunters is allowed in one hunting party.

- **3.8** Shotguns only are to be used in hunting pheasant. The use or possession of a rifle slug or ball is prohibited.
- **3.9** The *Wildlife Conservation Act* Fur Harvesting Regulations pertaining to the hunting of wildlife in the province shall also apply to the hunting on the Glenfinnan Wildlife Management Area.
- **3.10** Every person who violates or attempts to violate any provision of these regulations is guilty of an offence and is liable, on summary conviction, to the fines specified under section 32 of the Act.

INDIAN RIVER WILDLIFE MANAGEMENT AREA

4. The following area of land is designated as a wildlife management area:

DESIGNATION

COMMENCING at the point of the intersection of the eastern boundary of the Malpeque Road, leading from Kensington to Malpeque, and the extension of the southern boundary of the Murphy Road, said point having coordinates N.412873.8, E.335369.1:

THENCE in a westerly direction along the southern boundary of the Murphy Road for a distance of 2,200 feet more or less to a point, the said point being the northwest corner of property of Lowell Clark;

THENCE running southwardly along the western boundary of said property of Lowell Clark for a distance of 1,400 feet more or less or to the northern boundary of property of Wilbur MacArthur;

THENCE westerly along the said northern boundary of property of Wilbur MacArthur for a distance of 3,000 feet more or less or to the eastern boundary of property presently owned by Patricia Emdin.

THENCE northwesterly along the said eastern boundary of property of said Patricia Emdin for a distance of 790 feet to a point;

THENCE S.74 20'30"W on a line passing through a point have co-ordinates N.410659.44 and E.329780.18 for a distance of approximately 825 feet or to the western boundary of property of Patricia Emdin;

THENCE in a northwesterly direction along the said western boundary of property of said Patricia Emdin and the extension thereof, for a distance of approximately 280 feet or to a point on the western boundary of the Clermont-Indian River Road;

THENCE in a northerly and northwesterly direction following the said western boundary of the Clermont-Indian River Road to the point of intersection of the said western boundary of the Clermont-Indian River Road and the northern boundary of the Hamilton Road, said point having co-ordinates N.418511.7, E.329865.2;

THENCE in an easterly direction following the said northern boundary of the Hamilton Road to the point of intersection of the said northern boundary of the Hamilton Road the and eastern boundary of the Malpeque Road, said point having coordinates N.418644.9, E333404.7;

THENCE southwardly following the said eastern boundary of the Malpeque Road to the point of intersection of the said eastern boundary of the Malpeque Road and the southern boundary of the Murphy Road extension, the said point being the point of commencement.

ALL azimuths and co-ordinates are derived from the Prince Edward Island Stereographic Projection System with the co-ordinates being expressed in feet.

REGULATIONS

- **4.1** No person shall, within the boundaries of the Indian River Wildlife Management Area,
 - (a) discharge a firearm within the designated area;
 - (b) operate a boat powered by a gasoline, diesel or electric engine unless authorized to do so by the Director; or
 - (c) hunt, trap or otherwise molest wildlife, except that angling for sports fish is permitted, in accordance with the *Wildlife Conservation Act* Angling Regulations.
- **4.2** Every person who violates or attempts to violate any provision of these regulations is guilty of an offence and is liable, on summary conviction, to the fines specified under section 32 of the Act.

MONTAGUE WILDLIFE MANAGEMENT AREA

5. The following area of land is designated as a wildlife management area:

DESIGNATION

ALL of the waters of the Montague and Valleyfield Rivers being within the boundaries described as follows and containing within the described area 125 acres more or less, and as shown on the attached plan:

COMMENCING at the point of the ordinary highwater mark of the south side of the Montague River at the base of the dam at Knox Pond.

THENCE easterly and southerly following the various courses of the ordinary highwater marks of the Montague and Valleyfield Rivers to the southern boundary of Route 326 at Sutherlands Bridge.

THENCE eastwardly along the said southern boundary of said Route 326 to a point on the ordinary highwater mark on the east side of the Valleyfield River.

THENCE northerly and easterly following the various courses of the ordinary highwater marks of the Valleyfield and Montague Rivers until it is intersected by the projection northward of the boundary between parcel number 193920 now in possession of The Lobster Shanty North Limited, and parcel number 193516 now in possession of Robert Gordon Mills.

THENCE northwardly following the said projection northward to the ordinary highwater mark on the north side of the said Montague River.

THENCE westwardly following the various courses of the ordinary highwater mark to the base of the dam aforesaid at Knox Pond.

THENCE southerly along the said base of the dam to the place of commencement.

REGULATIONS

- **5.1** No person shall, within the boundaries of the Montague Wildlife Management Area,
 - (a) discharge a firearm;
 - (b) operate a canoe or boat powered by a gasoline or diesel engine upstream of the Main Street Bridge unless authorized to do so by the Minister of Fisheries and Aquaculture.
- **5.2** No person shall hunt, or otherwise molest or disturb wildlife in the Montague Wildlife Management Area provided that nothing in these regulations shall prohibit any person from angling sports fish, in accordance with the *Wildlife Conservation Act* Angling Regulations or from trapping furbearing animals, in accordance with the *Wildlife Conservation Act* Fur Harvesting Regulations.
- **5.3** Every person who violates or attempts to violate any provision of these regulations is guilty of an offence and is liable, on summary conviction, to the fines specified under section 32 of the Act.

MOORE'S WILDLIFE MANAGEMENT AREA

6. The following area of land is designated as a wildlife management area:

DESIGNATION

ALL THAT TRACT, PIECE OR PARCEL OF LAND situate, lying and being on Lots 59 and 61 in Kings County, Province of Prince Edward Island, bounded and described as follows:

COMMENCING at a point at the intersection of the northern boundary of the Whim Road and extension of the western boundary of land presently owned by Mrs. Lester MacDonald.

THENCE southeasterly along the said western boundary of Mrs. Lester MacDonald's land to the southwest corner thereof,

THENCE eastwardly along the southern boundary of the said Mrs. Lester MacDonald's land and in a line in continuation thereof to the northeast corner of property presently in possession of the Department of Fisheries, Aquaculture and Environment.

THENCE southeasterly along the eastern boundary of the said property in possession of the Department of Fisheries, Aquaculture and Environment and in a line in continuation thereof to the southern boundary of the Sturgeon Road,

THENCE southwesterly along the said southern boundary of the Sturgeon Road until it intersects with the extension of the western boundary of the property presently in possession of Claude Matheson,

THENCE northwesterly along the said western boundary of the property presently in possession of Claude Matheson to the northwest corner thereof,

THENCE eastwardly along the northern boundary of the property presently in possession of Claude Matheson and in a line in continuation thereof to the southwest corner of property presently in possession of the Land Development Corporation,

THENCE northwesterly along the western boundary of the said property of the Land Development Corporation and in a line in continuation thereof to the said northern boundary of the Whim Road,

THENCE eastwardly along the northern boundary of the Whim Road to the place of commencement.

CONTAINING in the described area 1184 acres.

THE ABOVE DESCRIBED area includes all the lands and waters within the described boundaries including all portions of the public roads included in the described area. (EC831/74)

REGULATIONS

- **6.1** No person shall, within the boundaries of Moore's Wildlife Management Area, hunt, kill, capture, take, injure or molest wildlife, or take, injure, destroy, or molest their nests or eggs, except as otherwise permitted, in accordance with the regulations.
- **6.2** No person shall without lawful excuse have in his or her possession any migratory bird or portion thereof or the nest or egg of any such bird while within the boundaries of Moore's Wildlife Management Area.
- **6.3** (1) No person shall have in his or her possession, while within the boundaries of Moore's Wildlife Management Area, any firearm or any decoy or other appliance of a kind used for hunting, killing, capturing or taking of migratory birds without being in possession of a valid and subsisting permit issued by the Director.
- (2) Subject to the provisions of the Act and the regulations made thereunder, the Director may, by permit, authorize in any year a person to have firearms in his or her possession and to shoot and have in his or her possession wild ducks and wild geese in such portion of Moore's Wildlife Management Area and during such time as the Minister may from time to time permit.
- **6.4** (1) No person shall keep a dog or cat in Moore's Wildlife Management Area or take or bring a dog or cat upon a bird sanctuary unless the person has a permit authorizing the person to do so, issued by the Director.
- (2) No owner of a dog or cat shall permit his or her dog or cat to run at large in Moore's Wildlife Management Area.
- (3) A game officer may destroy any dog or cat found chasing or molesting migratory birds in Moore's Wildlife Management Area.
- (4) A game office may seize a dog or cat the officer finds running at large in Moore's Wildlife Management Area.
- (5) A game officer who has seized a dog or cat under subsection (4) may, in the officer's discretion, restore possession of the dog or cat to the owner thereof where
 - (a) the owner claims possession of the dog or cat within five days after the date of seizure; and
 - (b) the owner pays to the game officer all expenses incurred in securing, caring for and feeding the dog or cat.
- (6) Where at the end of five days possession, if the dog or cat has not been restored to the owner under subsection (5) the game officer may sell the dog or cat by public auction.
- (7) The proceeds of the sale of a dog or cat by public auction shall be distributed in the following manner:
 - (a) all expenses incurred in securing, caring for and feeding the dog or cat shall be paid to the game officer;
 - (b) the expenses of the public auction shall be paid; and
 - (c) the balance, if any, shall be paid to the owner.
- (8) Where a dog or cat has not been reclaimed within five days after seizure under subsection (5) and no bid has been received at a sale by public auction, the game officer may destroy or dispose of the dog or cat as the officer sees fit.
- **6.5** No person shall carry on in Moore's Wildlife Management Area, any activity that is detrimental to migratory birds or their eggs or nests unless the person has a permit authorizing the person to do so, issued by the Minister.

- **6.6** No person shall operate a boat powered by a gasoline engine within the Moore's Wildlife Management Area.
- **6.7** No person shall hunt or trap or otherwise molest wildlife within the Moore's Wildlife Management Area.
- **6.8** Every person who violates or attempts to violate any provision of these regulations is guilty of an offence and is liable, on summary conviction, to the fines specified under section 32 of the Act.

NEW GLASGOW WILDLIFE MANAGEMENT AREA

7. The following area of land is designated as a wildlife management area:

DESIGNATION

ALL THAT TRACT, piece or parcel of land and water, lying and being at New Glasgow, Lot 23, in Queens County, P.E.I. containing approximately 438 acres, more particularly described as follows:

COMMENCING at a point on the western boundary of Rennies Road said point being on the southern boundary extended westwardly of a road running along the southern boundary of land now, or formerly, in the possession of Elmer MacDonald, (P.I.D. Number 230045);

THENCE eastwardly along the southern boundary of said road running along Elmer MacDonald's southern boundary to a point on the eastern boundary of the Mt. Tom Road;

THENCE northwardly along said eastern boundary of the Mt. Tom Road until it meets the northern boundary of the New Glasgow Road; THENCE northwestwardly along said eastern boundary of the New Glasgow Road to the western boundary of the Rennies Road;

AND THENCE southwardly along the western boundary of Rennies Road to the place of commencement.

REGULATIONS

- **7.1** No person shall discharge a firearm from or within the boundaries of the New Glasgow Wildlife Management Area.
- **7.2** No person shall hunt, or otherwise molest or disturb wildlife in the New Glasgow Wildlife Management Area provided that nothing in these regulations shall prohibit any person from angling sports fish, in accordance with the *Wildlife Conservation Act* Angling Regulations or from trapping furbearing animals, in accordance with the *Wildlife Conservation Act* Fur Harvesting Regulations.
- **7.3** Every person who violates or attempts to violate any provision of these regulations is guilty of an offence and is liable, on summary conviction, to the fines specified under section 32 of the Act.

ORWELL COVE WILDLIFE MANAGEMENT AREA

Revoked by EC457/93

PISOUID RIVER WILDLIFE MANAGEMENT AREA

8. The following area of land is designated as a wildlife management area:

DESIGNATION

ALL THAT PARCEL OF LAND situate, lying and being on Lot or Township No. 37 in Queens County, Province of Prince Edward Island, bounded and described as follows, that is to say:

COMMENCING at a point located at the intersection of the northern boundary of the Fort Augustus Road (Highway No. 21) and the western boundary of the Dromore Road, said point having coordinates East 309992, North 198360;

THENCE in an eastwardly direction along the northern boundary of the Fort Augustus Road to the point at the intersection of the said northern boundary of the Fort Augustus Road, the eastern boundary of Highway No. 22, and the southwestern angle of lands now or formerly in the possession of John and Leona Doran, said point having coordinates East 311055, North 198700;

THENCE in a southwardly direction along the eastern boundary of Highway No. 22 to a point located in the eastern boundary aforesaid and being 40 metres, a little more or less, in a southeastwardly direction from the southeastern angle of lands now or formerly in the possession of Leslie Mitchell, said point having coordinates East 312785, North 197320;

THENCE in a westwardly direction along the southern boundary of Joeys Road to a point located at the intersection of the southern boundary aforesaid, the western boundary of the Dromore Road, the northeastern angle of lands now or formerly in the possession of the Prince Edward Island Agricultural Development Corporation and

the southeastern angle of lands now or formerly in the possession of Bernadette Walsh, said point having coordinates East 311120, North 196690.

THENCE in a northwardly direction along the western boundary of the Dromore Road to the point at the place of commencement;

CONSISTING of approximately 232.5 hectares (574.5 acres) of land.

THE ABOVE DESCRIBED area includes all the lands and waters within the described boundaries including all portions of the public roads included in the described area.

ALL coordinates are derived from the Prince Edward Island Stereographic Projection System with the coordinates being expressed in metres.

REGULATIONS

- **8.1** No person shall discharge a firearm from or within the boundaries of the Pisquid River Wildlife Management Area.
- **8.2** No person shall hunt or trap or otherwise molest wildlife in the Pisquid River Wildlife Management Area provided that nothing in these regulations shall prohibit any person from angling sports fish, in accordance with the Maritime Fishery Regulations, or trapping furbearing animals, after the close of the migratory game bird hunting season in Prince Edward Island as prescribed in Schedule 1 of the *Migratory Birds Convention Act* Regulations.
- **8.3** Every person who violates or attempts to violate any provision of these regulations is guilty of an offence and is liable, on summary conviction, to the fines specified under section 32 of the Act.

ROLLO BAY WILDLIFE MANAGEMENT AREA

9. The following area of land is designated as a wildlife management area:

DESIGNATION

ALL THAT TRACT, PIECE OR PARCEL OF LAND situate, lying and being in Lot of Township Number 43 in Kings County, Province of Prince Edward Island, bounded and described as follows:

COMMENCING at a point situated on the southeastern corner of lands now or formerly in possession of Alphonse Peters:

THENCE on an azimuth of 164 17'51" for a distance of 187.21 meters to point number 3 having coordinates E351124.02 and N200642.08;

THENCE 178 02'35" for a distance of 93.80 meters to point number 4 having coordinates E351127.22 and N200548.33;

THENCE 141 21'01" for a distance of 36.82 meters to point number 5 having coordinates E351150.21 and N200519.58;

THENCE 189 43'32" for a distance of 28.21 meters to point number 6 having

coordinates E351145.45 and N200491.78;

THENCE 115 26'41" for a distance of 126.48 meters to point number 7 having

coordinates E351259.66 and N200437.44;

THENCE 193 45'09" for a distance of 152.38 meters to point number 8 situated on the northern shore of Rollo Bay; said point number 8 having coordinates E351223.43 and N200289.42;

THENCE 189 06'15" for a distance of 1368.40 meters to point number 9 situated at Rollo Point; said point number 9 having coordinates E351006.91 and N198938.26;

THENCE in a southwesterly direction along the various courses of the shore of Rollo Bay to point number 20 having coordinates E350399.35 and N198314.06;

THENCE 292 35'30" for a distance of 108.00 meters to point number 21 having coordinates E350299.64 and N198355.55;

THENCE 207 37'21" for a distance of 49.16 meters to point number 22 having coordinates E350276.84 and N198311.99;

THENCE 283 22'38" for a distance of 170 meters more or less to a point situated on the western boundary of the Rollo Bay Road;

THENCE in a northerly direction and northwesterly direction along the southwestern boundary of the Rollo Bay Road to Fortune Bridge;

THENCE in a northeasterly direction along the western boundary of Fortune Bridge Road to the northern boundary of Highway No. 2;

THENCE in an easterly direction along the northerly boundary of Highway No. 2 to intersect the northern prolongation of the eastern boundary of said lands of Alphonse Peters;

THENCE in a southerly direction along the eastern boundary of said lands of Alphonse Peters to the point at the place of commencement;

ALL AZIMUTHS and coordinates are derived from the Prince Edward Island Double Stereographic Projection System with the coordinates expressed in meters.

REGULATIONS

- **9.1** No person shall discharge a firearm from or within the boundaries of the Rollo Bay Wildlife Management Area.
- **9.2** No person shall hunt, or otherwise molest wildlife in the Rollo Bay Wildlife Management Area provided that nothing in these regulations shall prohibit any person from angling sports fish, in accordance with the *Wildlife Conservation Act* Angling Regulations.
- **9.3** Every person who violates or attempts to violate any provision of these regulations is guilty of an offence and is liable, on summary conviction, to the fines specified under section 32 of the Act.

SCHDULE B

DINGWELLS MILLS

- 1. The following area of land is designated as a wildlife management area:
- **ALL THOSE PARCELS OF LAND** situate, lying, and being on Lot or Township No. 56 in Kings County, Province of Prince Edward Island, described as follows, that is to say:

PARCEL 1

Includes all that portion of land identified as provincial property number 144576, lying West of the Selkirk Route, also known as Route 309, and being bound on the North by lands identified as provincial property number 144584 and now or formerly in the possession of Russell Rogers, On the East by the Selkirk Road, On the south by lands identified as provincial property number 483859 and now or formerly being in the possession of Eloi Leger, On the West by the Fortune River, consisting of approximately 14 acres of land, a little more or less.

PARCEL 2:

Includes all lands identified as provincial property number 144626, consisting of approximately 20.2 hectares (50.0 acres) of land, a little more or less, EXCEPTING THEREOUT AND THEREFROM all that parcel of land East of the Selkirk Road, also known as Route 309, and thus being a portion of the lands listed in a Deed of Conveyance dated the 15th day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 433 with the Registrar of Deeds for Kings County on the 15th day of March, A. D. 1994 in Book 286, Page 29.

PARCEL 3:

Includes all lands identified as provincial property number 165324, consisting of approximately 50.6 hectares (125.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 15th day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 433 with the Registrar of Deeds for Kings County on the 15th day of March, A. D. 1994 in Book 286, Page 29.

PARCEL 4:

Includes all lands identified as provincial property number 165332, consisting of approximately 101.2 hectares (250.0 acres) of land, a little more or less.

PARCEL 5:

Includes a portion of the lands as identified as provincial property number 165373, lying West of the Selkirk Road and being bounded on the East by the Selkirk Road and on the South by Route #2, also known as the Souris Road, consisting of approximately 114.5 acres of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 15th day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 433 with the Registrar of Deeds for Kings County on the 15th day of March, A. D. 1994 in Book 286, Page 29.

PARCEL 6:

Includes all lands identified as provincial property number 165456, consisting of approximately 25.1 hectares (62.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 25th day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 497 with the Registrar of Deeds for Kings County on the 29th day of March, A. D. 1994 in Book 286, Page 70.

PARCEL 7:

Includes all lands identified as provincial property number 165472, consisting of approximately 23.6 hectares (58.25 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 15th day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 433 with the Registrar of Deeds for Kings County on the 15th day of March, A. D. 1994 in Book 286, Page 29.

PARCEL 8:

Includes all lands identified as provincial property number 165506, being a portion of the lands as described in a Deed of Conveyance dated the 17th day of March, A. D. 1966 from Willard L. Jordan and Estelle Jordan to Her Majesty the Queen and registered as Document Number 170 with the Registrar of Deeds for Kings County on the 17th day of March, A. D. 1966 in Book 85, Page 348. Excepting and Reserving thereout and therefrom all that parcel of land as described in a Deed of Conveyance dated the 25th day of August, A.D. 1977 from The Province of Prince Edward to Roger Blaisdell and registered in the Kings County Registry Office on the 20th day of September, A.D. 1977 in Book 121, Page 17, as Document Number 1352; Also Excepting and Reserving thereout and therefrom all that parcel of land as described in a Deed of Rectification dated the 30th day of July, A.D. 1986 from The Government of Prince Edward Island to Gordon Blaisdell and registered in the Kings County Registry Office on the 21st day of August, A.D. 1986 in Book 194, Page 32, as Document Number 1277; Also Excepting and Reserving thereout and therefrom all that parcel of land as described in a Deed of Conveyance dated the 2nd day of January, A.D. 1998, from the Government of Prince Edward Island to Peter Peters and registered in the Kings County Registry Office on the 12th day of February, A.D. 1998 in Book 370, Page 22, as Document Number 275. The Above Described Lands containing by estimation approximately 438 acres of land a little more or less; and including all that portion of lands bearing same provincial property number 165506 and lying South of Highway Route #2

PARCEL 9:

Includes all lands identified as provincial property number 435123, consisting of approximately 13.6 hectares (33.5 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 26th day of April, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 698 with the Registrar of Deeds for Kings County on the 27th day of April, A. D. 1994 in Book 287, Page 89.

PARCEL 10:

Includes all lands identified as provincial property number 469189, consisting of approximately 10.1 hectares (25.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 15th day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 433 with the Registrar of Deeds for Kings County on the 15th day of March, A. D. 1994 in Book 286, Page 29.

PARCEL 11:

Includes all lands identified as provincial property number 543660, consisting of approximately 80.2 hectares (198.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 26th day of April, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 698 with the Registrar of Deeds for Kings County on the 27th day of April, A. D. 1994 in Book 287, Page 89.

PARCEL 12:

Includes all lands identified as provincial property number 823377, consisting of approximately 33.6 hectares (83.0 acres) of land, a little more or less, **EXCEPTING THEREOUT AND THEREFROM** all that parcel of land East of the Selkirk Road, also known as Route 309, and thus being a portion of the lands described in a Deed of Conveyance dated the 21st day of December, A. D. 1994 from the Prince Edward Island Agricultural Insurance Corporation and the Canada-Prince Edward Island Crop Insurance Fund to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 2441 with the Registrar of Deeds for Kings County on the 21st day of December, A. D. 1994 in Book 298, Page 69.

DROMORE

2. The following area of land is designated as a wildlife management area:

ALL THOSE PARCELS OF LAND situate, lying, and being on Lot or Township No. 37 in Queens County, Province of Prince Edward Island, described as follows, that is to say:

PARCEL 1:

Includes all lands identified as provincial property number 123026, consisting of approximately 30.4 hectares (75.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 17th day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 1450 with the Registrar of Deeds for Queens County on the 18th day of March, A.D. 1994 in Book 718, Page 55.

PARCEL 2:

Includes all lands identified as provincial property number 125179, consisting of approximately 22.7 hectares (56.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 17th day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 1450 with the Registrar of Deeds for Queens County on the 18th day of March, A.D. 1994 in Book 718, Page 55.

PARCEL 3:

Includes all lands identified as provincial property number 125195, consisting of approximately 12.1 hectares (30.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 17th day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 1450 with the Registrar of Deeds for Queens County on the 18th day of March, A.D. 1994 in Book 718, Page 55.

PARCEL 4:

Includes all lands identified as provincial property number 125211, consisting of approximately 42.9 hectares (106.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 17th day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 1450 with the Registrar of Deeds for Queens County on the 18th day of March, A.D. 1994 in Book 718, Page 55.

PARCEL 5:

Includes all lands identified as provincial property number 125369, consisting of approximately 1.6 hectares (4.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 25th day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 1618 with the Registrar of Deeds for Queens County on the 29th day of March, A.D. 1994 in Book 719, Page 19.

PARCEL 6:

Includes all lands identified as provincial property number 125393, consisting of approximately 17.4 hectares (43.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 17th day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 1450 with the Registrar of Deeds for Queens County on the 18th day of March, A.D. 1994 in Book 718, Page 55.

PARCEL 7:

Includes all lands identified as provincial property number 125476, consisting of approximately 4.9 hectares (12.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 17th day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 1450 with the Registrar of Deeds for Queens County on the 18th day of March, A.D. 1994 in Book 718, Page 55.

PARCEL 8:

Includes all lands identified as provincial property number 125575, consisting of approximately 15.0 hectares (37.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 17th day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 1450 with the Registrar of Deeds for Queens County on the 18th day of March, A.D. 1994 in Book 718, Page 55.

PARCEL 9:

Includes all lands identified as provincial property number 301770, consisting of approximately 51.4 hectares (127.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 17th day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 1450 with the Registrar of Deeds for Queens County on the 18th day of March, A.D. 1994 in Book 718, Page 55.

PARCEL 10:

Includes all lands identified as provincial property number 303891, consisting of approximately 140.7 hectares (347.5 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 17th day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 1450 with the Registrar of Deeds for Queens County on the 18th day of March, A.D. 1994 in Book 718, Page 55.

PARCEL 11:

Includes all lands identified as provincial property number 304519, consisting of approximately 21.1 hectares (52.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 17th day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 1450 with the Registrar of Deeds for Queens County on the 18th day of March, A.D. 1994 in Book 718, Page 55.

PARCEL 12:

Includes all lands identified as provincial property number 304816, consisting of approximately 16.2 hectares (40.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 17th day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 1450 with the Registrar of Deeds for Queens County on the 18th day of March, A.D. 1994 in Book 718, Page 55.

PARCEL 13:

Includes all lands identified as provincial property number 408880, consisting of approximately 12.1 hectares (30.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 17th day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 1450 with the Registrar of Deeds for Queens County on the 18th day of March, A.D. 1994 in Book 718, Page 55.

PARCEL 14:

Includes all lands identified as provincial property number 470716, consisting of approximately 1.6 hectares (4.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 7th day of July, A. D. 1966 from Maurice Corrigan and Wife to Her Majesty the Queen and registered as Document Number 139 with the Registrar of Deeds for Queens County on the 8th day of July, A.D. 1966 in Book 156, Page 671.

PARCEL 15:

Includes all lands identified as provincial property number 572875, consisting of approximately 9.3 hectares (23.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 18th day of February, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 1618 with the Registrar of Deeds for Queens County on the 21st day of February, A.D. 1994 in Book 716, Page 22.

PARCEL 16:

Includes all lands identified as provincial property number 635433, consisting of approximately 25.5 hectares (63.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 25th day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 1618 with the Registrar of Deeds for Queens County on the 29th day of March, A.D. 1994 in Book 719, Page 19.

PARCEL 17:

Includes all lands identified as provincial property number 703660, consisting of approximately 8.1 hectares (20.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 26th day of April, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 2189 with the Registrar of Deeds for Queens County on the 27th day of April, A.D. 1994 in Book 721, Page 44.

PARCEL 18:

Includes all lands identified as provincial property number 715326, consisting of approximately 7.2 hectares (17.7 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 17th day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 1450 with the Registrar of Deeds for Queens County on the 18th day of March, A.D. 1994 in Book 718, Page 55.

PARCEL 19:

ALL THAT PARCEL OF LAND situate, lying, and being on Lot or Township No. 38 in Kings County, Province of Prince Edward Island, described as follows, that is to say: Includes all lands identified as provincial property number 123034, consisting of approximately 19.4 hectares (48.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 17th day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 449 with the Registrar of Deeds for Kings County on the 18th day of March, A.D. 1994 in Book 286, Page 38.

GROVEPINE - BIG BROOK

3. The following area of land is designated as a wildlife management area:

ALL THOSE PARCELS OF LAND situate, lying, and being on Lot or Township No. 56 in Kings County, Province of Prince Edward Island, described as follows, that is to say:

PARCEL 1:

Includes all lands identified as provincial property number 152041 consisting of approximately 27.1 hectares (67.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 26th day of April, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 698 with the Registrar of Deeds for Kings County on the 27th day of April, A.D. 1994 in Book 287, Page 89.

PARCEL 2:

Includes all lands identified as provincial property number 152058 consisting of approximately 54.3 hectares (134.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 26th day of April, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 698 with the Registrar of Deeds for Kings County on the 27th day of April, A.D. 1994 in Book 287, Page 89.

PARCEL 3:

Includes all lands identified as provincial property number 152116 consisting of approximately 34.4 hectares (85.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 17th day of March, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 449 with the Registrar of Deeds for Kings County on the 18th day of March, A.D. 1994 in Book 286, Page 38.

PARCEL 4:

Includes all lands identified as provincial property number 152124 consisting of approximately 37.7 hectares (93.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 17th day of March, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 449 with the Registrar of Deeds for Kings County on the 18th day of March, A.D. 1994 in Book 286, Page 38.

PARCEL 5:

Includes all lands identified as provincial property number 152140, consisting of approximately 40.1 hectares (99.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 26th day of April, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 698 with the Registrar of Deeds for Kings County on the 27th day of April, A.D. 1994 in Book 287, Page 89. **EXCEPTING THEREOUT AND THEREFROM** all lands identified as provincial property number 540070, consisting of approximately 0.9 hectares (2.1 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 23rd day of June, A. D. 1998 from the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works to Bernard J. Cheverie and Stacey D. MacDonald and registered as Document Number 1218 with the Registrar of Deeds for Kings County on the 23rd day of June, A. D. 1998 in Book 379, Page 20.

PARCEL 6:

Includes all lands identified as provincial property number 152207 consisting of approximately 268.4 hectares (663.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 24th day of February, A.D. 1989 from the Souris Trading Company Ltd. to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 313 with the Registrar of Deeds for Kings County on the 27th day of February, A.D. 1989 in Book 223, Page 27.

PARCEL 7:

Includes all lands identified as provincial property number 152462 consisting of approximately 21.1 hectares (52.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 24th day of February, A.D. 1989 from the Souris Trading Company Ltd. to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 313 with the Registrar of Deeds for Kings County on the 27th day of February, A.D. 1989 in Book 223, Page 27.

PARCEL 8:

Includes all lands identified as provincial property number 152488, consisting of approximately 75.3 hectares (186.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 24th day of February, A.D. 1989 from the Souris Trading Company Ltd. to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 313 with the Registrar of Deeds for Kings County on the 27th day of February, A.D. 1989 in Book 223, Page 27.

PARCEL 9:

Includes all lands identified as provincial property number 152652, lying situate and being in Howe Bay, Lot or Township no. 56, Kings County, containing by estimation approximately 52.6 hectares (130 acres) of land, a little more or less.

PARCEL 10:

Includes all lands identified as provincial property number 153841 consisting of approximately 33 acres of land, a little more or less, and thus being described in a Deed of Conveyance dated the 13th day of April, A.D. 1999, from Garth Robertson to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document No. 769 with the Registrar of Deeds for Kings County on the 20th day of April, A.D. 1999, in Book 399, Page 30.

PARCEL 11:

Includes all lands identified as provincial property number 153874, consisting of approximately 35.2 hectares (87.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 25th day of March, A. D. 1994 from the Prince Edward Island Agricultural Insurance Corporation and the Canada-Prince Edward Island Crop Insurance Fund to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 497 with the Registrar of Deeds for Kings County on the 29th day of March, A. D. 1994 in Book 286, Page 70.

PARCEL 12:

Includes all lands identified as provincial property number 155564 consisting of approximately 64.8 hectares (160.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 24th day of February, A.D. 1989 from the Souris Trading Company Ltd. to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 313 with the Registrar of Deeds for Kings County on the 27th day of February, A.D. 1989 in Book 223, Page 27.

PARCEL 13:

Includes all lands identified as provincial property number 155580 consisting of approximately 36.4 hectares (90.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 17th day of March, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 449 with the Registrar of Deeds for Kings County on the 18th day of March, A.D. 1994 in Book 286, Page 38.

PARCEL 14:

Includes all lands identified as provincial property number 155598, consisting of approximately 81.0 hectares (200.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 20th day of April, A.D. 1966 from James O. Hornby to Her Majesty the Queen and registered as Document Number 242 with the Registrar of Deeds for Kings County on the 20th day of April, A.D. 1966 in Book 85, Page 415

PARCEL 15:

Includes all lands identified as provincial property number 155606, consisting of approximately 32.4 hectares (80.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 25th day of March, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 497 with the Registrar of Deeds for Kings County on the 29th day of March. A.D. 1994 in Book 286. Page 70.

PARCEL 16:

Includes all lands identified as provincial property number 155705 consisting of approximately 40.5 hectares (100.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 15th day of March, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 433 with the Registrar of Deeds for Kings County on the 15th day of March, A.D. 1994 in Book 286, Page 29.

PARCEL 17:

Includes all lands identified as provincial property number 165399 consisting of approximately 34.4 hectares (85.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 15th day of March, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 433 with the Registrar of Deeds for Kings County on the 15th day of March, A.D. 1994 in Book 286, Page 29.

PARCEL 18:

Includes all lands identified as provincial property number 165407 consisting of approximately 25.1 hectares (62.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 15th day of March, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 433 with the Registrar of Deeds for Kings County on the 15th day of March, A.D. 1994 in Book 286, Page 29.

PARCEL 19:

Includes all lands identified as provincial property number 165514 consisting of approximately 60.7 hectares (150.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 17th day of March, A.D. 1966 from Willard L. Jordan and Wife to Her Majesty the Queen and registered as Document Number 170 with the Registrar of Deeds for Kings County on the 17th day of March, A.D. 1966 in Book 85, Page 348.

PARCEL 20:

Includes all lands identified as provincial property number 425025 consisting of approximately 20.2 hectares (50.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 17th day of March, A.D. 1966 from Willard L. Jordan and Wife to Her Majesty the Queen and registered as Document Number 170 with the Registrar of Deeds for Kings County on the 17th day of March, A.D. 1966 in Book 85, Page 348.

PARCEL 21:

Includes all that portion of land identified as provincial property number 484394 lying West of the Grovepine Road, consisting of approximately 37 acres of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 26th day of April, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 698 with the Registrar of Deeds for Kings County on the 27th day of March, A.D. 1994 in Book 287, Page 89.

PARCEL 22:

Includes all lands identified as provincial property number 484691 consisting of approximately 4.0 hectares (10.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 24th day of February, A.D. 1989 from the Souris Trading Company Ltd. to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 313 with the Registrar of Deeds for Kings County on the 27th day of February, A.D. 1989 in Book 223, Page 27.

PARCEL 23:

Includes all lands identified as provincial property number 488544 consisting of approximately 23.5 hectares (58.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 8th day of February, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 216 with the Registrar of Deeds for Kings County on the 8th day of February, A.D. 1994 in Book 285, Page 4.

ALL THOSE PARCELS OF LAND situate, lying, and being on Lot or Township No. 55 in Kings County, Province of Prince Edward Island, described as follows, that is to say:

PARCEL 24:

Includes all lands identified as provincial property number 154260 consisting of approximately 43.3 hectares (107.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 23rd day of October, A.D. 1995 from Newsco Investments Ltd. to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 1963 with the Registrar of Deeds for Kings County on the 23rd day of October, A.D. 1995 in Book 317, Page 54.

PARCEL 25:

Includes all lands identified as provincial property number 154351 consisting of approximately 35.3 hectares (87.3 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 23rd day of October, A.D. 1995 from Newsco Investments Ltd. to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 1963 with the Registrar of Deeds for Kings County on the 23rd day of October, A.D. 1995 in Book 317, Page 54.

PARCEL 26:

Includes all lands identified as provincial property number 155549 consisting of approximately 19.5 hectares (48.2 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 23rd day of October, A.D. 1995 from Newsco Investments Ltd. to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 1963 with the Registrar of Deeds for Kings County on the 23rd day of October, A.D. 1995 in Book 317, Page 54

PARCEL 27:

Includes all lands identified as provincial property number 155648, consisting of approximately 56.7 hectares (140.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 17th day of March, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 449 with the Registrar of Deeds for Kings County on the 18th day of March, A.D. 1994 in Book 286, Page 38.

PARCEL 28:

Includes all lands identified as provincial property number 155671 consisting of approximately 23.6 hectares (58.3 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 23rd day of October, A.D. 1995 from Newsco Investments Ltd. to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 1963 with the Registrar of Deeds for Kings County on the 23rd day of October, A.D. 1995 in Book 317, Page 54.

ALL THAT PARCEL OF LAND situate, lying, and being on Lot or Township No. 43 in Kings County, Province of Prince Edward Island, described as follows, that is to say:

PARCEL 29:

Includes all lands identified as provincial property number 823625, consisting of approximately 24.3 hectares (60.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 13th day of January, A. D. 1995 from the Prince Edward Island Agricultural Insurance Corporation and the Canada-Prince Edward Island Crop Insurance Fund to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 78 with the Registrar of Deeds for Kings County on the 16th day of January, A. D. 1995 in Book 299, Page 41.

MARTINVALE - CORRAVILLE

4. The following area of land is designated as a wildlife management area:

ALL THOSE PARCELS OF LAND situate, lying, and being on Lot or Township No. 53 in Kings County, Province of Prince Edward Island, described as follows, that is to say:

PARCEL 1:

Includes all lands identified as provincial property number 156943, consisting of approximately 4.9 hectares (12.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 26th day of April, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 698 with the Registrar of Deeds for Kings County on the 27th day of April, A. D. 1994 in Book 287, Page 89.

PARCEL 2:

Includes all lands identified as provincial property number 161406, consisting of approximately 40.5 hectares (100.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 6th day of March, A.D. 1986 from David R. Jardine to the Government of Prince Edward Island as represented by the Minister of Transportation and Public works and registered as Document Number 310 with the Registrar of Deeds for Kings County on the 6th day of March, A. D. 1986 in Book 190, Page 16.

PARCEL 3:

Includes all lands identified as provincial property number 161414, consisting of approximately 27.1 hectares (67.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 15th day of March, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 433 with the Registrar of Deeds for Kings County on the 15th day of March, A. D. 1994 in Book 286, Page 29.

PARCEL 4:

Includes all lands identified as provincial property number 161430, consisting of approximately 8.5 hectares (21.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 26th day of April, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 698 with the Registrar of Deeds for Kings County on the 27th day of March, A. D. 1994 in Book 287, Page 89.

PARCEL 5:

Includes all that portion of lands identified as provincial property number 161448, and lying 27 chains, 50 links West of the Martinvale Road, also known as Route 321, consisting of approximately 31 acres of land, a little more or less, and thus being a portion of the lands listed in a Deed of Conveyance dated the 15th day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 430 with the Registrar of Deeds for Kings County on the 15th day of March, A. D. 1994 in Book 286, Page 26.

PARCEL 6:

Includes all lands identified as provincial property number 161588, consisting of approximately 60.7 hectares (150.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 17th day of March, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 449 with the Registrar of Deeds for Kings County on the 18th day of March, A. D. 1994 in Book 286, Page 38.

PARCEL 7:

Includes all lands identified as provincial property number 161612, consisting of approximately 24.3 hectares (60.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 17th day of March, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 449 with the Registrar of Deeds for Kings County on the 18th day of March, A. D. 1994 in Book 286, Page 38.

PARCEL 8:

Includes all lands identified as provincial property number 161620, consisting of approximately 20.2 hectares (50.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 15th day of March, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 433 with the Registrar of Deeds for Kings County on the 15th day of March, A. D. 1994 in Book 286, Page 29.

PARCEL 9:

COMMENCING at a point located in the Southern boundary of lands now or formerly in the possession of Leonard Kelly, also known as property number 161703, said point located 950 meters, a little more or less, East from the Eastern boundary of the Martinvale Road; THENCE East along the Southern boundary of lands aforesaid for the distance of 950 meters, a little more or less, or to a point located in the Western boundary of lands now or formerly in the possession of the Government of Prince Edward Island, also known as property number 161950; THENCE South and at right angles along the Western boundary of lands aforesaid for the distance of 107.5 meters, a little more or less, or to a point located in the Northern boundary of lands now or formerly in the possession of the Government of Prince Edward Island, also known as property number 161612; THENCE West and at right angles along the Northern boundary of lands aforesaid for the distance of 950 meters, a little more or less; **THENCE** North and at right angles for the distance of 107.5 meters, a little more or less, or to the point at the place of commencement. CONTAINING approximately 30 acres of land, a little more or less. BEING AND INTENDED TO BE the Eastern portion of the lands identified as provincial property number 161695 and thus being listed in a Deed of Conveyance dated the 17th day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 449 with the Registrar of Deeds for Kings County on the 18th day of March, A. D. 1994 in Book 286, Page 38.

PARCEL 10:

Includes all lands identified as provincial property number 161745, consisting of approximately 34.4 hectares (85.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 17th day of March, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public works and registered as Document Number 449 with the Registrar of Deeds for Kings County on the 18th day of March, A. D. 1994 in Book 286, Page 38.

PARCEL 11:

Includes all lands identified as provincial property number 161752, consisting of approximately 14.2 hectares (35.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 17th day of March, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 449 with the Registrar of Deeds for Kings County on the 18th day of March, A. D.1994 in Book 286, Page 38.

PARCEL 12:

Includes all lands identified as provincial property number 161836, consisting of approximately 101.2 hectares (250 acres) of land, a little more or less.

PARCEL 13:

Includes all lands identified as provincial property number 161877, consisting of approximately 20.2 hectares (50.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 27th day of April, A.D. 1994 from Leonard and Betty MacKenzie to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 703 with the Registrar of Deeds for Kings County on the 28th day of April, A. D. 1994 in Book 287, Page 92.

PARCEL 14:

Includes all lands identified as provincial property number 161984, consisting of approximately 135.6 hectares (335.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 15th day of March, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 433 with the Registrar of Deeds for Kings County on the 15th day of March, A. D. 1994 in Book 286, Page 29.

PARCEL 15:

Includes all lands identified as provincial property number 162016, consisting of approximately 49.4 hectares (122.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 26th day of April, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 698 with the Registrar of Deeds for Kings County on the 27th day of March, A. D. 1994 in Book 287, Page 89.

PARCEL 16:

Includes all lands identified as provincial property number 162032, consisting of approximately 64.8 hectares (160.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 15th day of March, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 432 with the Registrar of Deeds for Kings County on the 15th day of March, A. D. 1994 in Book 286, Page 28.

PARCEL 17:

Includes all lands identified as provincial property number 162073, consisting of approximately 46.6 hectares (115.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 15th day of March, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 433 with the Registrar of Deeds for Kings County on the 15th day of March, A. D. 1994 in Book 286, Page 29.

PARCEL 18:

Includes all lands identified as provincial property number 162107, consisting of approximately 83.0 hectares (205.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 15th day of March, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 433 with the Registrar of Deeds for Kings County on the 15th day of March, A. D. 1994 in Book 286, Page 29.

PARCEL 19:

Includes all lands identified as provincial property number 162214, containing approximately 20.2 hectares (50.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 1st day of February, A. D. 1972 from the Farm Credit Corporation to Her Majesty the Queen and registered as Document Number 565 with the Registrar of Deeds for Kings County on the 25th day of March, A. D. 1972 in Book 93, Page 594.

PARCEL 20:

Includes all lands identified as provincial property number 162230, consisting of approximately 80.9 hectares (200.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 15th day of March, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 433 with the Registrar of Deeds for Kings County on the 15th day of March, A. D. 1994 in Book 286, Page 29

PARCEL 21:

Includes all lands identified as provincial property number 162248, consisting of approximately 56.7 hectares (140.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 26th day of August, A.D. 1981 from J. O. Hornby Ltd. to the Prince Edward Island Land Development Corporation and registered as Document Number 1202 with the Registrar of Deeds for Kings County on the 28th day of August, A. D. 1981 in Book 152, Page 31.

PARCEL 22:

Includes all lands identified as provincial property number 162305, consisting of approximately 40.5 hectares (100.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 8th day of February, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 216 with the Registrar of Deeds for Kings County on the 8th day of February, A. D. 1994 in Book 285, Page 4.

PARCEL 23:

Includes all lands identified as provincial property number 416651, consisting of approximately 8.7 hectares (21.5 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 26th day of April, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 698 with the Registrar of Deeds for Kings County on the 27th day of April, A. D. 1994 in Book 287, Page 89.

PARCEL 24:

Includes all that portion of lands identified as provincial property number 436618, lying East of the Martinvale Road, also known as Route 321, and consisting of approximately 24.3 hectares (60.0 acres) of land, a little more or less, and thus being a portion of the lands as described in a Deed of Conveyance dated the 14th day of January, A. D. 1974 from Gordon Shaw to The Prince Edward Island Development Corporation and registered as Document Number 161 with the Registrar of Deeds for Kings County on the 29th day of January, A. D. 1974 in Book 101, Page 194.

PARCEL 25:

Includes all lands identified as provincial property number 479683, consisting of approximately 35.6 hectares (88.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 15th day of March, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 433 with the Registrar of Deeds for Kings County on the 15th day of March, A. D. 1994 in Book 286, Page 29.

PARCEL 26:

Includes all lands identified as provincial property number 501114, consisting of approximately 23.7 hectares (58.5 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 25th day of March, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 497 with the Registrar of Deeds for Kings County on the 29th day of March, A. D. 1994 in Book 286, Page 70.

PARCEL 27:

Includes all lands identified as provincial property number 659698, consisting of approximately 21.1 hectares (52.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 15th day of March, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 433 with the Registrar of Deeds for Kings County on the 15th day of March, A. D. 1994 in Book 286, Page 29.

PARCEL 28:

Includes all lands identified as provincial property number 745844, consisting of approximately 18.2 hectares (45.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 20th day of December, A.D. 1988 from Harlan Melville Robbins to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 2159 with the Registrar of Deeds for Kings County on the 20th day of December, A. D. 1988 in Book 221, Page 14.

PARCEL 29:

Includes all lands identified as provincial property number 835504, consisting of approximately 21.5 hectares (53.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 24th day of February, A.D. 1995 from Mary Louise Campbell to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 346 with the Registrar of Deeds for Kings County on the 2nd day of March, A. D. 1995 in Book 301, Page 51.

PARCEL 30:

Includes all lands identified as provincial property number 861112, consisting of approximately 15.4 hectares (38.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 7th day of July, A. D. 1998 from Miles Matheson and Ronald E. Matheson to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 1342 with the Registrar of Deeds for Kings County on the 8th day of July, A. D. 1998 in Book 380, Page 21.

RIVER WETLANDS

5. The following area of land is designated as a wildlife management area:

ALL THOSE PARCELS OF LAND situate, lying, and being on Lot. or Township No. 42 in Kings County, Province of Prince Edward Island, described as follows, that is to say:

PARCEL 1:

Includes all lands identified as provincial property number 122457, consisting of approximately 20.2 hectares (50.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 19th day of December, A. D. 1991 from George Paton to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 2070 with the Registrar of Deeds for Kings County on the 19th day of December, A. D. 1991 in Book 259, Page 12.

PARCEL 2:

Includes all that portion of lands identified as provincial property number 122523 lying East of the Albion Road, also known as Route 327, consisting of approximately 45 acres of land, a little more or less, and thus being a portion of the lands listed in a Deed of Conveyance dated the 26th day of April, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 698 with the Registrar of Deeds for Kings County on the 26th day of April, A. D. 1994 in Book 287, Page 89.

PARCEL 3:

Includes all lands identified as provincial property number 165274, consisting of approximately 19.0 hectares (47.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 26th day of April, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 698 with the Registrar of Deeds for Kings County on the 26th day of April, A. D. 1994 in Book 287, Page 89.

PARCEL 4:

Includes all lands identified as provincial property number 181123, consisting of approximately 22.2 hectares (54.9 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 23rd day of October, A. D. 1995 from Newsco Investments Ltd. to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 1963 with the Registrar of Deeds for Kings County on the 23rd day of October, A. D. 1995 in Book 317, Page 54.

PARCEL 5:

Includes all lands identified as provincial property number 770818, consisting of approximately 13.8 hectares (34.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 13th day of September, A. D. 1991 from John J. Sheehan to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 1456 with the Registrar of Deeds for Kings County on the 16th day of September, A. D. 1991 in Book 255, Page 72.

PARCEL 6:

Includes all lands identified as provincial property number 780536, consisting of approximately 15.0 hectares (37.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 26th day of March, A. D. 1991 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 452 with the Registrar of Deeds for Kings County on the 26th day of March, A. D. 1991 in Book 250, Page 43.

PARCEL 7:

Includes all lands identified as provincial property number 784009, consisting of approximately 16.0 hectares (39.5 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 12th day of August, A. D. 1991 from J. L. Francis Burge and Catherine Burge to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 1249 with the Registrar of Deeds for Kings County on the 12th day of August, A. D. 1991 in Book 254, Page 51.

PARCEL 8:

Includes all lands identified as provincial property number 803460, consisting of approximately 18.62 hectares (46 acres) of land, a little more or less, and thus being described in the following Deeds of Conveyance: Deed from St. Lawrence Corporation Limited to the Government of Prince Edward Island, dated the 15th day of July 1993, and registered in the Kings County Registry Office on the 19th day of July 1993, in Book 277, Page 7, as Document Number 1120, and containing 3.44 hectares (8.5 acres) of land, a little more or less. Deed from Martinus Rose to the Government of Prince Edward Island, dated the 27th day of January 1994, and registered in the Kings County Registry Office on the 31st day of January 1994, in Book 284, Page 60, as Document Number 163, and containing 5.06 hectares (12.5 acres) of land, a little more or less. Deed from Gerard and Mary P. Dwan to the Government of Prince Edward Island, dated the 18th day of December 2002, and registered in the Kings County Registry Office on the 31st day of December 2002, in Book 490, Page 33, as Document Number 3140, and containing 10.12 hectares (25 acres) of land, a little more or less.

PARCEL 9:

Includes all lands identified as provincial property number 873216, consisting of approximately 4.9 hectares (12.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 29th day of July, A. D. 1999 from Donald Flynn to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 1772 with the Registrar of Deeds for Kings County on the 7th day of September, A. D. 1999 in Book 408, Page 33.

PARCEL 10:

Includes all lands identified as provincial property number 938530, consisting of approximately 7.8 hectares (19.4 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 19^{th} day of October, A.D. 2004 from Elizabeth Walsh to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 2496, with the Registrar of Deeds for Kings County on the 20^{th} day of October, A.D. 2004.

PARCEL 11:

Includes all lands identified as provincial property number 926816, consisting of approximately 8 hectares (20.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 16th day of April, A.D. 2004 from Karl and Thelma Hess to the Government of Prince Edward Island as represented by the Minister of Transportation of Public Works and registered as Document Number 877, with the Registrar of Deeds for Kings County on the 26th day of April, A.D. 2004 in Book 516, Page 53.

PARCEL 12:

Includes all lands identified as provincial property number 943928, consisting of approximately 13.3 hectares (32.84 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the $2^{\rm nd}$ day of December, A.D. 2005 from Joseph and Wendy Anderson to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 2807, with the Registrar of Deeds for Kings County on the $6^{\rm th}$ day of December, A.D. 2005.

MT. STEWART

6. The following area of land is designated as a wildlife management area:

ALL THOSE PARCELS OF LAND situate, lying, and being on Lot or Township No. 38 in Kings County, described as follows, that is to say:

PARCEL 1:

Includes all lands identified as provincial property number 147751, consisting of approximately 10.5 hectares (26.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 15th day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 433 with the Registrar of deeds for Kings County on the 15th day of March, A. D. 1994 in Book 286, Page 29.

26 MARCH 2009

PARCEL 2:

Includes all lands identified as provincial property number 147868, consisting of approximately 12.1 hectares (30.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 19th day of December, A. D. 1992 from Orville MacDonald to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 2090 with the Registrar of deeds for Kings County on the 21st day of December, A. D. 1992 in Book 270, Page 97.

PARCEL 3:

Includes all lands identified as provincial property number 436592, consisting of approximately 5.7 hectares (14.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 15th day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 433 with the Registrar of deeds for Kings County on the 15th day of March, A. D. 1994 in Book 286, Page 29.

PARCEL 4:

Includes all lands identified as provincial property number 466367, consisting of approximately 1.2 hectares (3.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 22nd day of December, A. D. 1992 from Lottie Woods to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 10 with the Registrar of deeds for Kings County on the 5th day of January, A. D. 1993 in Book 271, Page 35.

PARCEL 5:

Includes all lands identified as provincial property number 468298, consisting of approximately 1.2 hectares (3.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 26th day of April, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 698 with the Registrar of deeds for Kings County on the 27th day of April, A. D. 1994 in Book 287, Page 89.

PARCEL 6:

Includes all lands identified as provincial property number 539387, consisting of approximately 1.8 hectares (4.5 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 29th day of January, A. D. 1993 from Sterling Gunn to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 177 with the Registrar of deeds for Kings County on the 1st day of February, A. D. 1993 in Book 272, Page 19.

PARCEL 7:

Includes all that portion of lands identified as provincial property number 786632, COMMENCING on the Western boundary of lands now or formerly in possession of Linda Roper Chowen and James Chowen at a point where said boundary is intersected by the north boundary of lands now or formerly in possession of Frederick Douglas Smith; THENCE Westwardly along the northern boundary of said Frederick Douglas Smith and continuing in a straight line to a point on the eastern boundary of lands now or formerly in the possession of Greg and Bernadette Ryan; THENCE northwardly along said Eastern boundary of said Ryan's to the shore of the Hillsborough River; THENCE Eastwardly along the various courses of River to a point where the said shore is intersected by the Western boundary of lands of Chowen aforesaid; THENCE southwardly along the Western boundary of lands of said Chowen to the point or place of commencement; said lands being herein described consisting of approximately 4.05 hectares (10.0 acres) of land, a little more or less. Said lands being a portion of the lands described in a Deed of Conveyance dated the 19th day of August, A. D. 1991 from Eric W. Smith to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 1294 with the Registrar of deeds for Kings County on the 19th day of August, A. D. 1991 in Book 254, Page 72.

PARCEL 8:

Includes all lands identified as provincial property number 788653, consisting of approximately 3.7 hectares (9.2 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 3rd day of March, A. D. 1992 from Cecil Allayene McAssey to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 311 with the Registrar of deeds for Kings County on the 3rd day of March, A. D. 1992 in Book 261, Page 35.

PARCEL 9:

Includes all that portion of lands lying south of CNR Rail lands identified as provincial property number 794768, consisting of approximately 19.51 hectares (48.20 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 16th day of July, A. D. 1992 from Earnest Farquarson to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 1054 with the Registrar of Deeds for Kings County on the 16th day of July, A. D. 1992 in Book 265, Page 37.

PARCEL 10:

Includes all lands identified as provincial property number 799213, consisting of approximately 0.4 hectares (1.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 19th day of December, A. D. 1992 from Orville MacDonald to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 2090 with the Registrar of deeds for Kings County on the 21st day of December, A. D. 1992 in Book 270, Page 97.

PARCEL 11:

Includes all lands identified as provincial property number 799221, consisting of approximately 4.9 hectares (12.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 14th day of December, A. D. 1992 from the Montreal Trust Company to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 2031 with the Registrar of deeds for Kings County on the 14th day of December, A. D. 1992 in Book 270, Page 70.

PARCEL 12:

Includes all lands identified as provincial property number 799239, consisting of approximately 9.7 hectares (24.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 21st day of December, A. D. 1992 from Evelyn Walsh to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 271 with the Registrar of deeds for Kings County on the 21st day of December, A. D. 1992 in Book 271, Page 1

PARCEL 13:

Includes all lands identified as provincial property number 799437, consisting of approximately 3.2 hectares (8.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 1st day of February, A. D. 1993 from Greg Ryan to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 272 with the Registrar of deeds for Kings County on the 1st day of February, A. D. 1993 in Book 272, Page 20.

ALL THOSE PARCELS OF LAND situate, lying, and being on Lot or Township No. 37 in Queens County, described as follows, that is to say:

PARCEL 14:

Includes all lands identified as provincial property number 587725, consisting of approximately 29.6 hectares (73.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 18th day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 899 with the Registrar of deeds for Queens County on the 21st day of February, A. D. 1994 in Book 716, Page 22.

PARCEL 15:

Includes all lands identified as provincial property number 778472, consisting of approximately 38.9 hectares (96.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 27th day of March, A. D. 1991 from George Douglas Hayes to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 1452 with the Registrar of deeds for Queens County on the 27th day of March, A. D. 1991 in Book 623, Page 3.

PARCEL 16:

Includes all lands identified as provincial property number 784702, consisting of approximately 28.85 hectares (71.3 acres) of land, a little more or less, and thus being described in the following Deeds of Conveyance: Deed from Doris Isabel McAssey (Executrix of Estate of Wendell Ralph McAssey) to the Government of Prince Edward Island, dated the 10th day of September 1991, and registered in the Queens County Registry Office on the 11th day of September 1991, in Book 638, Page 46, as Document no. 6006, and containing 0.61 hectares (1.5 acres) of land a little more or less. Deed from Doris Isabel McAssey to the Government of Prince Edward Island, dated the 3rd day of December 1991, and registered in the Queens County Registry Office on the 4th day of December 1991, in Book 646, Page 92, as Document no. 8068, and containing 1.53 hectares (3.8 acres) of land a little more or less. Deed from Bruce Pigot (Dorothy Pigot and Frank Pigot as "life tenants") to the Government of Prince Edward Island, dated the 2nd day of November 1991, and registered in the Queens County Registry Office on the 12th day of November 1991, in Book 644, Page 65, as Document no. 7530, and containing 25.50 hectares (63.0 acres) of land a little more or less. Deed from Lloyd Doucette to the Government of Prince Edward Island, dated the 7th day of February 1992, and registered in the Queens County Registry Office on the 7th day of February 1992, in Book 651, Page 80, as Document no. 715, and containing 1.21 hectares (3.0 acres) of land a little more or less.

PARCEL 17:

Includes all lands identified as provincial property number 858720, consisting of approximately 9.5 hectares (23.5 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 18th day of March, A. D. 1998 from the Roman Catholic Episcopal Corporation of the Diocese of Charlottetown to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 1517 with the Registrar of Deeds for Queens County on the 19th day of March, A. D. 1998 in Book 925, Page 39.

PARCEL 18:

Includes all lands identified as provincial property number 683474, consisting of approximately 2.4 hectares (6.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 10th day of September, A. D. 1999 from Donald C. Steele to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 6123 with the Registrar of Deeds for Queens County on the 14th day of September, A. D. 1999 in Book 1017, Page 3.

PARCEL 19:

Includes all lands identified as provincial property number 874818, consisting of approximately 3.9 hectares (9.7 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 18th day of October, A. D. 1999 from Hudson Sanderson to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 7340 with the Registrar of Deeds for Queens County on the 29th day of October, A. D. 1999 in Book 1026, Page 42.

SOUTHAMPTON

7. The following area of land is designated as a wildlife management area:

ALL THOSE PARCELS OF LAND situate, lying, and being on Lot or Township Nos. 40 and 41 in Kings County, Province of Prince Edward Island, described as follows, that is to say:

PARCEL 1:

Includes all lands identified as provincial property number 631465, consisting of approximately 58.0 hectares (143.3 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 26th day of April, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 698 with the Registrar of Deeds for Kings County on the 27th day of April, A.D. 1994 in Book 287, Page 89.

PARCEL 2:

Includes all lands identified as provincial property number 156356, consisting of approximately 5.8 hectares (14.2 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 26th day of April, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 698 with the Registrar of Deeds for Kings County on the 27th day of April, A.D. 1994 in Book 287, Page 89.

PARCEL 3:

Includes all lands identified as provincial property number 156422, consisting of approximately 13.4 hectares (33.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 26th day of April, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 698 with the Registrar of Deeds for Kings County on the 27th day of April, A.D. 1994 in Book 287, Page 89.

PARCEL 4:

Includes all lands identified as provincial property number 120444, consisting of approximately 20.2 hectares (50.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 15th day of March, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 433 with the Registrar of Deeds for Kings County on the 15th day of March A.D. 1994 in Book 286, Page 29.

PARCEL 5:

Includes all lands identified as provincial property number 120469, consisting of approximately 28.3 hectares (70.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 15th day of March, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 433 with the Registrar of Deeds for Kings County on the 15th day of March, A.D. 1994 in Book 286, Page 29.

PARCEL 6:

Includes all lands identified as provincial property number 120428, consisting of approximately 25.9 hectares (64.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 15th day of March, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 433 with the Registrar of Deeds for Kings County on the 15th day of March, A.D. 1994 in Book 286, Page 29.

PARCEL 7:

Includes all lands identified as provincial property number 188748, consisting of approximately 9.3 hectares (23.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 15th day of March, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 433 with the Registrar of Deeds for Kings County on the 15th day of March, A.D. 1994 in Book 286, Page 29.

PARCEL 8:

Includes all lands identified as provincial property number 120501, consisting of approximately 27.5 hectares (68.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 15th day of March, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 433 with the Registrar of Deeds for Kings County on the 15th day of March, A.D. 1994 in Book 286, Page 29.

PARCEL 9:

Includes all lands identified as provincial property number 499772, consisting of approximately 23.5 hectares (58.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 15th day of March, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 433 with the Registrar of Deeds for Kings County on the 15th day of March, A.D. 1994 in Book 286, Page 29.

PARCEL 10:

Includes all lands identified as provincial property number 156182, consisting of approximately 39.7 hectares (98.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 19th day of August, A.D. 1970 from John Claude Matheson and Wife to Her Majesty the Queen and registered as Document Number 750 with the Registrar of Deeds for Kings County on the 21st day of August, A.D. 1970 in Book 90, Page 567.

PARCEL 11:

Includes all lands identified as provincial property number 120519, consisting of approximately 17.0 hectares (42.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 1st day of June, A.D. 1970 from Alvin Garrett and Wife to Her Majesty the Queen and registered as Document Number 739 with the Registrar of Deeds for Kings County on the 19th day of August, A.D. 1970 in Book 90, Page 559.

PARCEL 12:

Includes all lands identified as provincial property number 401281, consisting of approximately 49.4 hectares (122.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 24th day of August, A.D. 1970 from Francis MacKinnon to Her Majesty the Queen and registered as Document Number 762 with the Registrar of Deeds for Kings County on the 25th day of August, A.D. 1970 in Book 90, Page 577.

PARCEL 13:

Includes all lands identified as provincial property number 156695, consisting of approximately 28.3 hectares (70.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 6th day of February, A.D. 1970 from Pius A. MacDonald and Wife to Her Majesty the Queen and registered as Document Number 93 with the Registrar of Deeds for Kings County on the 6th day of February, A.D. 1970 in Book 90, Page 38.

PARCEL 14:

Includes all lands identified as provincial property number 188946, consisting of approximately 23.9 hectares (59.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 7th day of February, A.D. 1972 from Albert James to Her Majesty the Queen and registered as Document Number 181 with the Registrar of Deeds for Kings County on the 17th day of February, A.D. 1972 in Book 93, Page 314.

PARCEL 15:

Includes all lands identified as provincial property number 119834, consisting of approximately 27.1 hectares (67.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 4th day of May, A.D. 1972 from Bruce James and Wife to Her Majesty the Queen and registered as Document Number 469 with the Registrar of Deeds for Kings County on the 8th day of May, A.D. 1972 in Book 93, Page 534.

PARCEL 16:

Includes all lands identified as provincial property number 120527, consisting of approximately 17.8 hectares (44.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 20th day of October, A.D. 1970 from Neil MacIntyre and Wife to Her Majesty the Queen and registered as Document Number 40 with the Registrar of Deeds for Kings County on the 13th day of January, A.D. 1971 in Book 91, Page 320.

PARCEL 17:

Includes all lands identified as provincial property number 156703, consisting of approximately 12.2 hectares (30.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 10th day of May, A.D. 1965 from Lillian MacDonald to Her Majesty the Queen and registered as Document Number 238 with the Registrar of Deeds for Kings County on the 14th day of May, A.D. 1965 in Book 84, Page 285

PARCEL 18:

Includes all lands identified as provincial property number 120667, consisting of approximately 20.2 hectares (50.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 24th day of January, A.D. 1963 from Mary Steele and Ano. to Her Majesty the Queen and registered as Document Number 78 with the Registrar of Deeds for Kings County on the 10th day of February, A.D. 1964 in Book 83, Page 185.

PARCEL 19:

Includes all lands identified as provincial property number 188938, consisting of approximately 20.2 hectares (50.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 15th day of March, A.D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 430 with the Registrar of Deeds for Kings County on the 15th day of March, A.D. 1994 in Book 286, Page 26.

PARCEL 20:

Includes all lands identified as provincial property number 188813, consisting of approximately 21.1 hectares (52.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 3rd day of November, A.D. 1971 from Anastasia Wilson to the Prince Edward Island Land Development Corporation and registered as Document Number 1446 with the Registrar of Deeds for Kings County on the 9th day of November, A.D. 1971 in Book 92, Page 693.

PARCEL 21:

Includes all lands identified as provincial property number 120436, consisting of approximately 10.1 hectares (25.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 26th day of June, A.D. 1996 from Joseph R. MacKinnon and Barbara MacKinnon to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 1213 with the Registrar of Deeds for Kings County on the 3rd day of July, A.D. 1996 in Book 333, Page 51.

PARCEL 22:

Includes all lands identified as provincial property number 188805, consisting of approximately 7.3 hectares (18.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 14th day of April, A. D. 1972 from Bruce James and Noreen James to Her Majesty the Queen and registered as Document Number 399 with the Registrar of Deeds for Kings County on the 17th day of April, A. D. 1972 in Book 93, Page 485.

PARCEL 23:

Includes all lands identified as provincial property number 156711, consisting of approximately 0.8 hectares (2.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 3rd day of April, A.D. 1970 from James Reginald MacKenzie to Her Majesty the Queen and registered as Document Number 235 with the Registrar of Deeds for Kings County on the 6th day of April, A.D. 1970 in Book 90, Page 148.

PARCEL 24:

Includes all lands identified as provincial property number 650697, consisting of approximately 31.1 hectares (78.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 26th day of February, A.D. 1997 from Martinus Rose to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 453 with the registrar of Deeds for Kings County on the 4th day of March, A.D. 1997 in Book 348, Page

PARCEL 25:

Includes all lands identified as provincial property number 120634, consisting of approximately 10.1 hectares (25.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 6th day of December, A.D. 1997 from Marie MacDonald and Murdock Blaxland and Marie MacDonald, Murdock Blaxland, George MacDonald, Wallace MacDonald, Wayne MacDonald, Freda Garrett, and Phyllis MacKinnon to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 164 with the Registrar of Deeds for Kings County on the 29th day of January, A.D. 1997 in Book 369, Page 32.

PARCEL 26:

Includes all lands identified as provincial property number 856096, consisting of approximately 25.7 hectares (63.5 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 24th day of November, A.D. 1997 from Garrett Farms Ltd. to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and registered as Document Number 2223 with the Registrar of Deeds for Kings County on the 25th day of November, A.D. 1997 in Book 365, Page 44.

ST. CHRYSOSTOME

8. The following area of land is designated as a wildlife management area:

ALL THOSE PARCELS OF LAND situate, lying, and being on Lot or Township No. 15 in Prince County, Province of Prince Edward Island, described as follows, that is to say:

PARCEL 1:

Includes all lands identified as provincial property number 23689, consisting of approximately 4.9 hectares (12.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 1st day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and Registered as Document Number 786 with the Registrar of Deeds for Prince County on the 2nd day of March, A. D. 1994 in Book 585, Page 58.

PARCEL 2:

Includes all lands identified as provincial property number 23697, consisting of approximately 10.1 hectares (25.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 1st day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and Registered as Document Number 786 with the Registrar of Deeds for Prince County on the 2nd day of March, A. D. 1994 in Book 585, Page 58.

PARCEL 3:

Includes all lands identified as provincial property number 23838, consisting of approximately 12.1 hectares (30.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 1st day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and Registered as Document Number 3476 with the Registrar of Deeds for Prince County on the 2nd day of March, A. D. 1994 in Book 154, Page 88.

PARCEL 4:

Includes all lands identified as provincial property number 23879, consisting of approximately 20.2 hectares (50.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 1st day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and Registered as Document Number 786 with the Registrar of Deeds for Prince County on the 2nd day of March, A. D. 1994 in Book 585, Page 58.

PARCEL 5:

Includes all lands identified as provincial property number 24075, consisting of approximately 4.1 hectares (10.0 acres) of land, a little more or less, and thus being described in a Deed of Conveyance dated the 27th day of April, A. D. 1972 from Melanie Arsenault and Alins Arsenault to the Prince Edward Island Land Development Corporation and Registered as Document Number 1153 with the Registrar of Deeds for Prince County on the 9th day of June, A. D. 1972 in Book 147, Page 471.

PARCEL 6:

Includes all lands identified as provincial property number 24133, consisting of approximately 10.1 hectares (25.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 25th day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and Registered as Document Number 1132 with the Registrar of Deeds for Prince County on the 29th day of March, A. D. 1994 in Book 587, Page 54.

PARCEL 7:

Includes all lands identified as provincial property number 266007, consisting of approximately 10.1 hectares (25.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 1st day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and Registered as Document Number 786 with the Registrar of Deeds for Prince County on the 2nd day of March, A. D. 1994 in Book 585, Page 58.

PARCEL 8:

Includes all lands identified as provincial property number 266098, consisting of approximately 19.4 hectares (48.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 1st day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and Registered as Document Number 786 with the Registrar of Deeds for Prince County on the 2nd day of March, A. D. 1994 in Book 585, Page 58.

PARCEL 9:

Includes all lands identified as provincial property number 266106, consisting of approximately 18.9 hectares (46.8 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 1st day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and Registered as Document Number 786 with the Registrar of Deeds for Prince County on the 2nd day of March, A. D. 1994 in Book 585, Page 58.

PARCEL 10:

Includes all lands identified as provincial property number 266130, consisting of approximately 32.47 hectares (80.23 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 1st day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and Registered as Document Number 786 with the Registrar of Deeds for Prince County on the 2nd day of March, A. D. 1994 in Book 585, Page 58. **EXCEPTING THEREOUT AND THEREFROM**, ALL THAT PARCEL OF LAND situate, lying and being at St. Chrysostome, Lot or Township No. 15, in Prince County, Province of Prince Edward Island, bounded and described as follows, that is to say: **COMMENCING** at a point situate on the West boundary of Arsenault Mill Road, as the same is shown on a plan of survey prepared by Locus Surveys Ltd., entitled "Plan of Survey Showing Lot 06-1 & Parcel A, Being a Subdivision of Lands of Government of P.E.I., at St. Chrysostome" and certified by James A. Clow, P.E.I.L.S. on January 16, 2006, as Drawing No. 05416, said point being designated by legal survey marker no. 2763 having coordinates Easting 222606.306 and Northing 433596.421; **THENCE** on an azimuth of 191° 43' 32" for the distance of 65.81 feet to calculated point no. 2764; **THENCE** on an azimuth of 189° 15' 40" for the distance

of 34.19 feet to survey marker no. 2765; **THENCE** on an azimuth of 269° 20' 35" for the distance of 335.10 feet to survey marker no. 2761; **THENCE** on an azimuth of 3° 40' 13" for the distance of 100.00 feet to legal survey marker no. 2762; **THENCE** on an azimuth of 89° 37' 59" for the distance of 347.56 feet to legal survey marker no. 2763 being the point at the place of commencement.

BEING and intended to be **Parcel A** as shown on attached sketch, containing an area of 0.77 acres of land, a little more or less, to be appended to land of the Grantee, Marcel Arsenault (PID #420133).

ALSO EXCEPTING THEREOUT AND THEREFROM, ALL THAT PARCEL OF LAND situate, lying and being at St. Chrysostome, Lot or Township No. 15, in Prince County, Province of Prince Edward Island, bounded and described as follows, that is to say: COMMENCING at a point situate on the North boundary of Route No. 11, as the same is shown on a plan of survey prepared by Locus Surveys Ltd. entitled "Plan of Survey Showing Lot 06-1 & Parcel A, Being a Subdivision of Lands of Government of P.E.I, at St. Chrysostome" and certified by James A. Clow, P.E.I.L.S. on January 16, 2006, as Drawing No. 05416, said point being designated by legal survey marker no. 2770 having coordinates Easting 222937.875 and Northing 433432.518; THENCE on an azimuth of 264° 46' 02" for the distance of 300.00 feet to legal survey marker no. 2766; THENCE on an azimuth of 11° 43' 32" for the distance of 183.68 feet to legal survey marker no. 2768; THENCE on an azimuth of 84° 45' 15" for the distance of 300.00 feet to legal survey marker no. 2769; THENCE on an azimuth of 190° 45' 57" for the distance of 300.00 feet to legal survey marker no. 2769; THENCE on an azimuth of 190° 45' 57" for the distance of 300.00 feet to legal survey marker no. 2770 being the point at the place of commencement.

BEING and intended to be **Lot 06-1** as shown on the attached sketch, containing an area of 2.00 acres of land, a little more or less.

AZIMUTHS and coordinates being referenced to the Prince Edward Island Stereographic Projection Coordinate System, distance and coordinates being expressed in feet.

PARCEL 11:

Includes all lands identified as provincial property number 266197, consisting of approximately 20.2 hectares (50.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 1st day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and Registered as Document Number 786 with the Registrar of Deeds for Prince County on the 2nd day of March, A. D. 1994 in Book 585, Page 58.

PARCEL 12:

Includes all lands identified as provincial property number 266718, consisting of approximately 6.1 hectares (15.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 1st day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and Registered as Document Number 786 with the Registrar of Deeds for Prince County on the 2nd day of March, A. D. 1994 in Book 585, Page 58.

PARCEL 13:

Includes all lands identified as provincial property number 266726, consisting of approximately 27.5 hectares (68.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 1st day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and Registered as Document Number 786 with the Registrar of Deeds for Prince County on the 2nd day of March, A. D. 1994 in Book 585, Page 58.

PARCEL 14:

Includes all lands identified as provincial property number 266734, consisting of approximately 22.3 hectares (55.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 1st day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and Registered as Document Number 786 with the Registrar of Deeds for Prince County on the 2nd day of March, A. D. 1994 in Book 585, Page 58.

PARCEL 15:

Includes all lands identified as provincial property number 432005, consisting of approximately 3.6 hectares (9.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 1st day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and Registered as Document Number 786 with the Registrar of Deeds for Prince County on the 2nd day of March, A. D. 1994 in Book 585, Page 58.

PARCEL 16:

Includes all lands identified as provincial property number 446385, consisting of approximately 13.4 hectares (33.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 25th day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and Registered as Document Number 1129 with the Registrar of Deeds for Prince County on the 29th day of March, A. D. 1994 in Book 587, Page 51.

PARCEL 17:

Includes all lands identified as provincial property number 475046, consisting of approximately 16.2 hectares (40.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 25th day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and Registered as Document Number 1129 with the Registrar of Deeds for Prince County on the 29th day of March, A. D. 1994 in Book 587, Page 51.

PARCEL 18:

Includes all lands identified as provincial property number 498006, consisting of approximately 110.5 hectares (273.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 26th day of April, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and Registered as Document Number 1554 with the Registrar of Deeds for Prince County on the 28th day of April, A. D. 1994 in Book 590, Page 16.

EXCEPTING THEREOUT AND THEREFROM, all that parcel, piece of tract lying and being by estimation, approximately 105 metres south from the Southeastern boundary of lands now or formerly in the possession of Kenneth and Wendy Walker, Thence continuing in a southwardly direction along the Western boundary of the Higgins Road a distance of 123 metres to a point; Thence in a Westerly direction a distance of 105 metres to a point; Thence in an ontherly direction and parallel to the Higgins Road a distance of 123 metres; Thence in an Easterly direction a distance of 105 metres or until it reaches a point on the Western boundary of the Higgins Road. Containing by estimation 1.29 hectares (3.19 acres) of land a little more or less. Said lands being herein described containing by estimation 109.21 hectares (269.81 acres).

PARCEL 19:

Includes all lands identified as provincial property number 591461, consisting of approximately 17.0 hectares (42.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 25th day of March, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and Registered as Document Number 1129 with the Registrar of Deeds for Prince County on the 29th day of March, A. D. 1994 in Book 587, Page 51.

PARCEL 20:

Includes all lands identified as provincial property number 627968, consisting of approximately 12.5 hectares (31.0 acres) of land, a little more or less, and thus being listed in a Deed of Conveyance dated the 26th day of April, A. D. 1994 from the Prince Edward Island Agricultural Development Corporation to the Government of Prince Edward Island as represented by the Minister of Transportation and Public Works and Registered as Document Number 1554 with the Registrar of Deeds for Prince County on the 28th day of April, A. D. 1994 in Book 590, Page 16.

- **2.** (1) The *Wildlife Conservation Act* Wildlife Management Areas Revocation Regulations (EC225/00) are revoked.
- (2) The Fish and Game Protection Act Wildlife Management Area Idem Regulations (EC1203/66) are revoked.
- **3.** These regulations come into force on April 4, 2009.

Commencement

EXPLANATORY NOTES

The amendment revokes the Wildlife Management Area Regulations that were made pursuant to the former Fish and Game Protection Act, which is repealed. The revoked regulations are substituted and are added to the Wildlife Management Areas Regulations made pursuant to the Wildlife Conservation Act, which replaces the former Fish and Game Protection Act. What were once two sets of regulations dealing with wildlife management areas made under two separate statutes, are now joined under the current and appropriate Wildlife Conservation Act, which remains in force.